# Stationary Markov Perfect Equilibria in Discounted Stochastic Games<sup>\*</sup>

Wei  $He^{\dagger}$  Yeneng  $Sun^{\ddagger}$ 

This version: August 20, 2016

#### Abstract

The existence of stationary Markov perfect equilibria in stochastic games is shown under a general condition called "(decomposable) coarser transition kernels". This result covers various earlier existence results on correlated equilibria, noisy stochastic games, stochastic games with finite actions and stateindependent transitions, and stochastic games with mixtures of constant transition kernels as special cases. A remarkably simple proof is provided via establishing a new connection between stochastic games and conditional expectations of correspondences. New applications of stochastic games are presented as illustrative examples, including stochastic games with endogenous shocks and a stochastic dynamic oligopoly model.

Keywords: Stochastic game, stationary Markov perfect equilibrium, equi-

librium existence, (decomposable) coarser transition kernel, endogenous shocks,

dynamic oligopoly.

<sup>\*</sup>The authors are grateful to Rabah Amir, Darrell Duffie, Matthew Jackson, Jiangtao Li, Xiang Sun, Yifei Sun, Satoru Takahashi and Yongchao Zhang for helpful discussions. This work was presented in Institute of Economics, Academia Sinica, Taipei, November 2013; 5th Shanghai Microeconomics Workshop, June 2014; Asian Meeting of the Econometric Society, Taipei, June 2014; China Meeting of the Econometric Society, Xiamen, June 2014; 14th SAET Meeting, Tokyo, August 2014; and the Conference in honor of Abraham Neyman, Jerusalem, June 2015. This research was supported in part by the Singapore Ministry of Education Academic Research Fund Tier 1 grants R-122-000-159-112 and R-146-000-170-112. This version owes substantially to the careful reading and expository suggestions of a coeditor, an associate editor and two referees.

<sup>&</sup>lt;sup>†</sup>Department of Economics, Chinese University of Hong Kong, Hong Kong. E-mail: he.wei2126@gmail.com

<sup>&</sup>lt;sup>‡</sup>Department of Economics, National University of Singapore, 1 Arts Link, Singapore 117570. Email: ynsun@nus.edu.sg

# Contents

| 1             | Inti                        | roduction   | 3  |  |
|---------------|-----------------------------|---|----|--|
| 2             | Discounted Stochastic Games |   | 7  |  |
| 3 Main Result |                             | in Result   | 11 |  |
| 4             | Ap                          | olications  | 13 |  |
|               | 4.1                         | Stochastic games with endogenous shocks             | 13 |  |
|               | 4.2                         | Stochastic dynamic oligopoly with price competition | 16 |  |
| 5             | An                          | Extension   | 19 |  |
| 6             | Dis                         | cussion   | 20 |  |
| 7             | Concluding Remarks          |   | 25 |  |
| 8             | Ap                          | pendix  | 26 |  |
|               | 8.1                         | Proof of Claim 1                                    | 26 |  |
|               | 8.2                         | Proof of Theorem 1                                  | 27 |  |
|               | 8.3                         | Proof of Theorem 2                                  | 30 |  |
|               | 8.4                         | Proof of Proposition 2                              | 32 |  |
|               | 8.5                         | Proof of Proposition 3                              | 33 |  |
|               | 8.6                         | Proof of Propositions 4 and 5                       | 35 |  |
| R             | References                  |   |    |  |

### 1 Introduction

The class of stochastic games enriches the model of repeated games by allowing the stage games to vary with some publicly observable states. It has considerably widened the applications of repeated games.<sup>1</sup> In particular, a stochastic game is played in discrete time by a finite set of players and the past history is observable by all the players. At the beginning of each stage, Nature draws some state randomly. After the state is realized, the players choose actions and each player receives a stage payoff that depends on the current state and the actions. The game then moves to the next stage and a new random state is drawn, whose distribution depends on the previous state and chosen actions. The procedure is repeated at the new state. Each player's total payoff is the discounted sum of the stage payoffs.

A strategy for a player in a stochastic game is a mapping from the space of history to the space of mixed feasible actions. However, the so-called Markov strategies, which only depend on the current state instead of the entire past history, have received much attention in the literature. As noted in [31], the concept of Markov perfect equilibrium, which requires the players to use only Markov strategies, embodies various practical virtues and philosophical considerations, including conceptual and computational simplicity. Given that the relevant parameters in a stochastic game are time-independent, it is natural to require the Markov strategies to be time-independent as well. Equilibria based on such strategies are called stationary Markov perfect equilibria. In a stationary Markov perfect equilibrium, any two subgames with the same payoffs and action spaces will be played exactly in the same way. So "bygones" are really "bygones"; i.e., the past history does not matter at all.

Beginning with [42], the existence of stationary Markov perfect equilibria in discounted stochastic games remains an important problem. Existence results on such equilibria in stochastic games with compact action spaces and finite/countable state spaces have been established in a number of early papers.<sup>2</sup> Given the

<sup>&</sup>lt;sup>1</sup>See, for example, the book [34], and the recent survey chapters [24] and [25].

<sup>&</sup>lt;sup>2</sup>The result for two-player zero-sum games with finite actions and states was established in [42]. The existence result for the general case with finite actions and states was proved in [15], [41] and [43]. The case with finite state spaces and compact action spaces was shown in [44]. The result in [42] was

wide applications of stochastic games with general state spaces in various areas of economics, the existence of equilibria in stationary strategies for such games has been extensively studied in the last two decades. However, no general existence result, except for several special classes of stochastic games, has been obtained in the literature so far.<sup>3</sup>

The existence of correlated stationary Markov perfect equilibria for stochastic games was proved in [10] and [38].<sup>4</sup> These papers assumed that there is a randomization device publicly known to all players which is irrelevant to the fundamental parameters of the game. Stationary Markov perfect equilibria have been shown to exist in [11], [35] and [40] for stochastic games with some special structures on the state transitions. The paper [40] focused on stochastic games with finite actions and state-independent transitions. A class of stochastic games has been studied in [35], where the transition probability has a fixed countable set of atoms and its atomless part is a finite combination of atomless measures that do not depend on states and actions. Stochastic games with a specific product structure, namely stochastic games with noise, were considered in [11]. In particular, the noise variable in [11] is a history-irrelevant component of the state and could influence the payoff functions and transitions. Recently, it was shown in [27] and [28] that a stochastic game satisfying the usual conditions may not have any stationary Markov perfect equilibrium.<sup>5</sup> This implies that a general existence result could only hold under some suitable conditions.

The first contribution of this paper is to introduce a general condition to guarantee the existence of stationary Markov perfect equilibria in stochastic games.

extended by [39] to two-player nonzero-sum games with finite action spaces and countable state spaces. The existence result for countable state spaces and uncountable action spaces was then proved by [14] and [45]. For more detailed discussions about the literature on stochastic games and their applications, see the surveys [4], [24, 25].

<sup>&</sup>lt;sup>3</sup>Stochastic games possessing strategic complementarities were studied in [2, 3], [5], [9] and [37]; see [6] and [23] for further discussions. A related class of stochastic games in which the interaction between different players is sufficiently weak was studied in [21]. The existence of stationary *p*-equilibria for two-player games with finite actions and strong separability conditions on both stage payoffs and state transitions was proved in [19]. The existence of stationary Markov perfect equilibria in intergenerational stochastic games was considered in [22] and [36]. All these papers impose special conditions on the payoff functions and state transitions.

<sup>&</sup>lt;sup>4</sup>Additional ergodic properties were obtained in [10] under stronger conditions.

<sup>&</sup>lt;sup>5</sup>An error in the relevant example of [27] was pointed out in [28], and a new example was presented therein.

Based on this condition, we unify various existence results in the literature as discussed in the previous paragraph,<sup>6</sup> and also provide a new class of stochastic games that is useful for economic applications and cannot be handled by existing results. Our second contribution is methodological. We establish for the first time a connection between the equilibrium payoff correspondences in stochastic games and a general result on the conditional expectations of correspondences, and hence are able to provide a simple proof for the existence of stationary Markov perfect equilibria. In the following paragraphs, we will discuss our condition and results in details.

The transition probability in a general stochastic game is defined in terms of the actions and state in the previous period. As discussed above, it is assumed in [10], [11], [38] and [40] that the actions and state in the previous period do not enter the transition of the sunspot/noise/shock component of the states. As a result, a key component of the transition probability in those papers is not influenced by the actions and state in the previous period. However, dynamic economic models with random shocks are common, and it may seem natural to assume that the transition of shocks can endogenously depend on some important factors (actions/states) in the previous stage. As shown by the counterexample in [28], without any restriction on the transition kernel,<sup>7</sup> a stationary Markov perfect equilibrium may not exist. We introduce a new model called "stochastic games with endogenous shocks",<sup>8</sup> which allows the distribution of current period's shocks to directly depend on the "discrete" components of the states and actions from the previous period.<sup>9</sup>

In order to prove an equilibrium existence result in general stochastic games that cover stochastic games with endogenous shocks and the class of games in which the state transitions have a decomposable feature (see, for example, [35] and [40]), we propose a condition called "decomposable coarser transition kernels" in the

<sup>&</sup>lt;sup>6</sup>All those papers work with general payoffs, but impose special structure on the state transitions. <sup>7</sup>Here the transition kernel means the Radon-Nikodym derivative of the transition probability with respect to some reference measure on the state space.

<sup>&</sup>lt;sup>8</sup>In contrast to the sunspot idea in [10] and [38], the innovation of [11] is to allow the noise to be part of the stage payoffs. This feature is also shared by stochastic games with endogenous shocks as considered in this paper.

<sup>&</sup>lt;sup>9</sup>Dynamic economic models with some discrete components of actions and states are common, such as entry or exit by firms in [13] and [20].

sense that the transition kernel is decomposed as a sum of finitely many components with each component being the product of a "coarser" history-relevant transition function and a history-irrelevant density function. For the case of stochastic games with endogenous shocks, the history-irrelevant density functions represent the shocks. In particular, each discrete component of the states and actions from the previous period naturally contributes one component in the transition kernel, which describes the history dependence of the distribution of the shock. It thus captures the intuition that the distribution of the random shock can directly depend on the actions and states in the previous period.

Our Theorem 1 shows that under the condition of a decomposable coarser transition kernel, a stochastic game has a stationary Markov perfect equilibrium. A very simple proof is given by providing a new link between a convexity type result of [12] on the conditional expectation of a correspondence and the equilibrium existence problem in stochastic games. As a corollary to Theorem 1, we know that a stationary Markov perfect equilibrium exists in a stochastic game with endogenous shocks. Theorem 2 extends Theorem 1 by including an atomic part in the transition probability, and covers the main existence result in [35] as a special case. As an illustrative application of stochastic games with endogenous shocks, we consider a stochastic version of the dynamic oligopoly model as studied in [29, 30].<sup>10</sup> Note that our results on stochastic games with endogenous shocks and the stochastic dynamic oligopoly model cannot be covered by existing results in the relevant literature, because the transition of the shock component in the state explicitly depends on the parameters in the previous stage.

To study a dynamic problem with a stationary structure, the standard approach is to work with a reduced problem with a recursive structure, in which players' payoffs are given by a convex combination of the stage payoffs and players' continuation values in terms of the Bellman equations. To solve the existence problem for stochastic games, one often needs to work with a one-shot auxiliary game parameterized by the state and the continuation value function.<sup>11</sup> The central difficulty in the existence argument is typically due to the failure of the convexity

 $<sup>^{10}</sup>$ Further applications will be discussed in Remark 2.

<sup>&</sup>lt;sup>11</sup>For details, see Subsection 8.2.

of the correspondence R, which is the collection of measurable selections (in terms of the state variable) from the equilibrium payoff correspondence of the auxiliary game. As a result, the desirable closedness and upper hemicontinuity properties would fail for R, and hence the classical fixed-point argument is not applicable. To handle such issues, the standard approach is to work with the convex hull co(R)of R. The key insight of our proof is that this imposed convexity restriction is relaxed by showing the equivalence of the conditional expectations of co(R) and Runder the condition of a decomposable coarser transition kernel. The minimality of our condition for this "one-shot game" approach is also demonstrated from a technical point of view in the sense that the conditions are shown to be tight as in Propositions 4 and 5.

The rest of the paper is organized as follows. Section 2 presents a general model of discounted stochastic games. The main result is given in Section 3. Stochastic games with endogenous shocks and a stochastic dynamic oligopoly model are considered in Section 4. Section 5 provides an extension of the main existence result by allowing the transition kernel to have an atomic part. Section 6 discusses the relationship between our existence theorems and several related results in the literature, and also demonstrates the minimality of our condition. Section 7 concludes the paper. The Appendix collects the proofs.

## 2 Discounted Stochastic Games

An *m*-person discounted stochastic game can be defined in terms of (1) a state space, (2) a state-dependent feasible action correspondence for each player, (3) a stage-payoff for each player that depends on the state and action profile, (4) a discount factor for each player, and (5) a transition probability that depends on the state and action profile. Formally, an *m*-person discounted stochastic game is described as follows:

- $I = \{1, \cdots, m\}$  is the set of players.
- $(S, \mathcal{S})$  is a measurable space representing the states of nature, where  $\mathcal{S}$  is

countably generated.<sup>12</sup>

- For each  $i \in I$ ,  $X_i$  is player i' action space, which is a nonempty compact metric space endowed with its Borel  $\sigma$ -algebra  $\mathcal{B}(X_i)$ . Let  $X = \prod_{1 \le i \le m} X_i$ and  $\mathcal{B}(X) = \bigotimes_{1 \le i \le m} \mathcal{B}(X_i)$ . Then X is the space of all possible action profiles.
- For each  $i \in I$ , the set of feasible actions of player i at state s is given by  $A_i(s)$ , where  $A_i$  is an S-measurable,<sup>13</sup> nonempty and compact valued correspondence from S to  $X_i$ . Let  $A(s) = \prod_{i \in I} A_i(s)$  for each  $s \in S$ .
- For each  $i \in I$ ,  $u_i : S \times X \to \mathbb{R}$  is player *i*'s stage-payoff with an absolute bound *C* (i.e., for every  $i \in I$  and  $(s, x) \in S \times X$ ,  $|u_i(s, x)| \leq C$  for some positive real number *C*) such that  $u_i(s, x)$  is *S*-measurable in *s* for each  $x \in X$ and continuous in *x* for each  $s \in S$ .
- $\beta_i \in [0, 1)$  is player *i*'s discount factor.
- The law of motion for the states is given by the transition probability Q:  $S \times X \times S \rightarrow [0, 1]$ .<sup>14</sup> That is, if s is the state at stage t and  $x \in X$  is the action profile chosen simultaneously by the m players at this stage, then Q(E|s, x) is the probability that the state at stage t + 1 belongs to the set E given s and x.
  - 1.  $Q(\cdot|s, x)$  (abbreviated as  $Q_{(s,x)}$ ) is a probability measure on (S, S) for all  $s \in S$  and  $x \in X$ , and for all  $E \in S$ ,  $Q(E|\cdot, \cdot)$  is  $S \otimes \mathcal{B}(X)$ -measurable.
  - 2. For all s and x,  $Q(\cdot|s, x)$  is absolutely continuous with respect to a probability measure  $\lambda$  on (S, S). Let q be an  $S \otimes S \otimes \mathcal{B}(X)$ -measurable function from  $S \times S \times X$  to  $\mathbb{R}_+$  such that for any  $s \in S$  and  $x \in X$ ,

<sup>&</sup>lt;sup>12</sup>It means that there is a countable subset  $\mathcal{D}$  of  $\mathcal{S}$  such that  $\mathcal{S}$  is generated by the sets in  $\mathcal{D}$ . This condition is widely adopted. For example, the state spaces in [10], [11], and [27] are respectively assumed to be a compact metric space, a Polish space, a Borel subset of a Polish space, and hence are all countably generated.

<sup>&</sup>lt;sup>13</sup>The correspondence  $A_i$  is said to be *S*-measurable (resp. weakly *S*-measurable) if for any closed (resp. open) subset *B* of  $X_i$ , the set  $\{s \in S : A_i(s) \cap B \neq \emptyset\}$  belongs to *S*; see Definition 18.1 of [1]. In the literature, there are some papers which assume that the correspondence  $A_i$  is weakly measurable; see, for example, [11] and [38]. These two measurability notions coincide in our setting since  $A_i$  is compact valued; see Lemma 18.2 in [1].

<sup>&</sup>lt;sup>14</sup>Note that the payoff  $u_i$  and the transition probability Q only need to be defined on the graph of A. For simplicity, we follow the literature to define them on the whole product space  $S \times X$ , as in [10], [27], [28] and [38].

 $q(\cdot, s, x)$  (also written as  $q(\cdot|s, x)$  or  $q_{(s,x)}$ ) is the corresponding Radon-Nikodym derivative of  $Q(\cdot|s, x)$ .

For all s ∈ S, the mapping x → Q(·|s, x) is norm continuous; that is, for any sequence of action profiles {x<sup>n</sup>} converging to some x<sup>0</sup>, the sequence {Q(·|s, x<sup>n</sup>)} converges to Q(·|s, x<sup>0</sup>) in total variation.<sup>15</sup>

A strategy of player *i* is a measurable mapping  $f_i$  from the past history to the set of player *i*'s mixed actions  $\mathcal{M}(X_i)$ ,<sup>16</sup> which places probability 1 on the set of feasible actions  $A_i(s)$  at each state *s*. For any profile of strategies  $f = \{f_i\}_{i \in I}$  of the players and every initial state  $s_1 \in S$ , a probability measure  $P_{s_1}^f$  is defined on  $(S \times X)^\infty$  in a canonical way; see, for example, [7]. Given the strategy profile *f* in the game starting from the state  $s_1$ , the expected payoff of player *i* is  $\mathbb{E}_{s_1}^f \left(\sum_{t=1}^\infty \beta_i^{t-1} u_i(s_t, x_t)\right)$ , where  $s_t$  and  $x_t$  describe the state and the action profile in stage *t*, and the expectation is taken with respect to the probability measure  $P_{s_1}^f$ . If a strategy does not depends on the history, then it is called a Markov strategy. As highlighted in [31], Markov strategies have the advantage that these strategies only depend on payoff-relevant data in the game.

In this paper, we shall focus on a particular class of Markov strategies, namely the "stationary Markov strategies", in which a player makes his decision based only on the current state but not the calendar time. Stationary Markov strategies are natural for the discounted payoff evaluation, since each subgame starting at the same state are strategically equivalent in the sense that players have the same payoffs in these subgames. In addition, stationary Markov strategies are particularly useful because they are easy to compute and implement. Formally, a stationary Markov strategy for player i is an S-measurable mapping  $f_i : S \to$  $\mathcal{M}(X_i)$  such that  $f_i(s)$  places probability 1 on the set  $A_i(s)$  for each  $s \in S$ .<sup>17</sup> A stationary Markov strategy profile f is called a **stationary Markov perfect** 

<sup>&</sup>lt;sup>15</sup>The total variation distance of two probability measures  $\mu$  and  $\nu$  on  $(S, \mathcal{S})$  is  $\|\mu - \nu\|_{TV} = \sup_{D \in \mathcal{F}} |\mu(D) - \nu(D)|$ . A sequence of probability measures  $\{\mu_n\}$  is said to be convergent to a probability measure  $\mu_0$  in total variation if  $\lim_{n\to\infty} \|\mu_n - \mu_0\|_{TV} \to 0$ .

<sup>&</sup>lt;sup>16</sup>For a Borel set D in a complete separable metric space, let  $\mathcal{M}(D)$  be the space of all Borel probability measures on D.

<sup>&</sup>lt;sup>17</sup>Since  $A_i$  is measurable, nonempty and compact valued, the correspondence  $A_i$  has a measurable graph by Theorem 18.6 in [1]. Then by Corollary 18.27 in [1],  $A_i$  has a measurable selection. As a result, the set of stationary Markov strategies of player *i* is nonempty for each *i*.

equilibrium if

$$\mathbb{E}_{s_1}^f\left(\sum_{t=1}^\infty \beta_i^{t-1} u_i(s_t, x_t)\right) \ge \mathbb{E}_{s_1}^{(g_i, f_{-i})}\left(\sum_{t=1}^\infty \beta_i^{t-1} u_i(s_t, x_t)\right)$$

for any  $i \in I$ ,  $s_1 \in S$  and any strategy  $g_i$  of player i.

In the following, we shall consider stationary Markov perfect equilibria in terms of the recursive structure, which is much easier to work with. By standard results in dynamic programming and stochastic games (see, for example, [8] and [38]), this formulation is equivalent to the equilibrium notion defined above. Given a stationary Markov strategy profile f, the continuation value  $v(\cdot, f)$  gives an essentially bounded S-measurable mapping from S to  $\mathbb{R}^m$ , which is uniquely determined by the following recursion

$$v_i(s,f) = \int_X \left[ (1 - \beta_i) u_i(s,x) + \beta_i \int_S v_i(s_1,f) Q(\mathrm{d}s_1|s,x) \right] f(\mathrm{d}x|s).^{18}$$
(1)

The strategy profile f is a **stationary Markov perfect equilibrium** if the discounted expected payoff of each player i is maximized by her strategy  $f_i$  in every state  $s \in S$ . It means that the continuation value v solves the following recursive maximization problem:

$$v_{i}(s, f) = \max_{x_{i} \in A_{i}(s)} \int_{X_{-i}} \left[ (1 - \beta_{i}) u_{i}(s, x_{i}, x_{-i}) + \beta_{i} \int_{S} v_{i}(s_{1}, f) Q(\mathrm{d}s_{1}|s, x_{i}, x_{-i}) \right] f_{-i}(\mathrm{d}x_{-i}|s),$$
(2)

where  $x_{-i}$  and  $X_{-i}$  have the usual meanings, and  $f_{-i}(s)$  is the product probability  $\otimes_{j\neq i} f_j(s)$  on the product of the action spaces of all players other than player *i* at the state *s*.

<sup>&</sup>lt;sup>18</sup>The existence and uniqueness of the continuation value of each player follows from a standard contraction mapping argument, see [8].

### 3 Main Result

In this section, we show the existence of stationary Markov perfect equilibria for discounted stochastic games. In particular, we introduce the notion of "(decomposable) coarser transition kernel" and present our main result. Subsection 8.2 provides a simple proof via establishing a new connection between stochastic games and conditional expectations of correspondences.

Before moving to the statement of the condition and the result, we need a formal concept of an atom over a sub- $\sigma$ -algebra. Suppose that the probability measure  $\lambda$ on the measurable state space (S, S) is atomless. Let  $\mathcal{G}$  be a sub- $\sigma$ -algebra of S. A set  $D \in S$  of positive measure is said to be a  $\mathcal{G}$ -atom if the restricted  $\sigma$ -algebras of  $\mathcal{G}$  and S to D are essentially the same. When the relevant  $\sigma$ -algebras are used to represent information, an event  $D \in S$  is a  $\mathcal{G}$ -atom simply means that given the realization of event D, S and  $\mathcal{G}$  carry the same information. Formally, let  $\mathcal{G}^D$  and  $\mathcal{S}^D$  be the respective  $\sigma$ -algebras  $\{D \cap D' : D' \in \mathcal{G}\}$  and  $\{D \cap D' : D' \in S\}$  on D. The set  $D \in S$  is said to be a  $\mathcal{G}$ -atom<sup>19</sup> if the strong completion of  $\mathcal{G}^D$  is  $\mathcal{S}^D$ .<sup>20</sup>

**Definition 1.** A discounted stochastic game is said to have a coarser transition kernel if for some sub- $\sigma$ -algebra  $\mathcal{G}$  of  $\mathcal{S}$ ,  $q(\cdot|s, x)$  is  $\mathcal{G}$ -measurable for all  $s \in S$  and  $x \in X$ , and  $\mathcal{S}$  has no  $\mathcal{G}$ -atom.<sup>21</sup>

A discounted stochastic game is said to have a decomposable coarser transition kernel if for some sub- $\sigma$ -algebra  $\mathcal{G}$  of  $\mathcal{S}$ ,  $\mathcal{S}$  has no  $\mathcal{G}$ -atom and for some positive integer J,  $q(s_1|s, x) = \sum_{1 \leq j \leq J} q_j(s_1, s, x) \rho_j(s_1)$ , where  $q_j$  is  $\mathcal{S} \otimes \mathcal{S} \otimes \mathcal{B}(X)$ -jointly measurable and  $q_j(\cdot, s, x)$  is  $\mathcal{G}$ -measurable for each  $s \in S$  and  $x \in X$ ,  $q_j(\cdot, s, x)$ and  $\rho_j$  are nonnegative and integrable on the atomless probability space  $(S, \mathcal{S}, \lambda)$ ,  $j = 1, \ldots, J$ .

Note that the condition of "coarser transition kernel" is a special case of the condition of "decomposable coarser transition kernel".

<sup>&</sup>lt;sup>19</sup>The notion of a  $\mathcal{G}$ -atom has been considered in [17] and [33], see also Definition 1.3.3 in [26]. The authors of this paper are grateful to an anonymous referee for providing these references.

<sup>&</sup>lt;sup>20</sup>The strong completion of  $\mathcal{G}^D$  in  $\mathcal{S}$  under  $\lambda$  is the set of all sets in the form  $E \triangle E_0$ , where  $E \in \mathcal{G}^D$ and  $E_0$  is a  $\lambda$ -null set in  $\mathcal{S}^D$ , and  $E \triangle E_0$  denotes the symmetric difference  $(E \setminus E_0) \cup (E_0 \setminus E)$ .

<sup>&</sup>lt;sup>21</sup>When  $\mathcal{G}$  is the trivial  $\sigma$ -algebra  $\{S, \emptyset\}$ ,  $\mathcal{S}$  has no  $\mathcal{G}$ -atom if and only if  $\lambda$  is atomless.

Below, we provide a simple example to illustrate the condition of "a decomposable coarser transition kernel".

**Example 1.** Suppose that the state space is  $S = [0,1] \times [0,1]$  (resp.  $S_1 = [0,1] \times [0,1]$ ) with the generic element s = (h,r) (resp.  $s_1 = (h_1,r_1)$ ),  $\lambda$  is the uniform distribution on the unit square, and the space of action profiles is  $X \subseteq \mathbb{R}^l$ . For simplicity, the density function constructed below does not depend on  $x \in X$ .

Let

$$\rho_1(s_1) = h_1, \qquad \rho_2(s_1) = r_1;$$

$$q_1(s_1, s) = \frac{h_1 + h}{2/3 + h}, \qquad q_2(s_1, s) = \frac{2h_1 + r}{1 + r};$$

$$q(s_1|s) = q(s_1, s) = q_1(s_1, s)\rho_1(s_1) + q_2(s_1, s)\rho_2(s_1).$$

It is easy to check that the integration of  $q(s_1, s)$  with respect to  $s_1$  on  $S_1$  is 1 for any  $s \in S$ .

Let S be the Borel  $\sigma$ -algebra on  $[0,1] \times [0,1]$  and  $\mathcal{G} = \mathcal{B}([0,1]) \otimes \{\emptyset, [0,1]\}$ , where  $\mathcal{B}([0,1])$  is the Borel  $\sigma$ -algebra on [0,1]. For any  $s \in S$ , both  $q_1(\cdot,s)$  and  $q_2(\cdot,s)$  are linear functions of  $h_1$  and do not depend on  $r_1$ . Thus, the  $\sigma$ -algebra generated by  $\{q_1(\cdot,s), q_2(\cdot,s)\}_{s\in S}$  is  $\mathcal{G}.^{22}$  As a result, the density function q satisfies the condition of a decomposable coarser transition kernel.

Notice that the functions  $\{q(\cdot, s)\}_{s \in S}$  are measurable with respect to the  $\sigma$ algebra S. Below, we show that S is indeed generated by the collection of functions  $\{q(\cdot, s)\}_{s \in S}$ . The proof is left in the Appendix.

**Claim 1.** In Example 1, the  $\sigma$ -algebra generated by  $\{q(\cdot, s)\}_{s \in S}$  is S.

The measurability requirements on the action correspondences, the stage payoffs, the transition probability and the transition kernel simply mean that the  $\sigma$ -algebra S is generated by the four collections of mappings  $\{A_i(\cdot)\}_{i \in I}$ ,  $\{u_i(\cdot, x)\}_{i \in I, x \in X}, \{Q(E|\cdot, x)\}_{E \in S, x \in X} \text{ and } \{q(\cdot|s, x)\}_{s \in S, x \in X}$ . When the transition kernel is in the form  $q(s_1|s, x) = \sum_{1 \leq j \leq J} q_j(s_1, s, x) \rho_j(s_1)$ , let  $\mathcal{G}$  be the  $\sigma$ algebra generated by the mappings  $\{q_j(\cdot|s, x)\}_{j \in J, s \in S, x \in X}$ . In the context of

<sup>&</sup>lt;sup>22</sup>A  $\sigma$ -algebra is said to be generated by a collection of mappings taking values in some complete separable metric spaces if it is the smallest  $\sigma$ -algebra on which these mappings are measurable.

Example 1, if one works with some S-measurable action correspondences and stage payoffs, then the mappings  $\{A_i(\cdot)\}_{i\in I}$ ,  $\{u_i(\cdot, x)\}_{i\in I, x\in X}$ ,  $\{Q(E|\cdot, x)\}_{E\in S, x\in X}$  and  $\{q(\cdot|s, x)\}_{s\in S, x\in X}$  generate the  $\sigma$ -algebra  $S = \mathcal{B}([0, 1]) \otimes \mathcal{B}([0, 1])$  (which is already generated by  $\{q(\cdot|s, x)\}_{s\in S, x\in X}$ , as shown in Claim 1). On the other hand, The  $\sigma$ algebra generated by  $\{q_1(\cdot, s), q_2(\cdot, s)\}_{s\in S}$  is  $\mathcal{G} = \mathcal{B}([0, 1]) \otimes \{\emptyset, [0, 1]\}$ . The condition of a decomposable coarser transition kernel is satisfied in this case because S has no  $\mathcal{G}$ -atom.

We now state the main result of this paper.

**Theorem 1.** Every discounted stochastic game with a (decomposable) coarser transition kernel has a stationary Markov perfect equilibrium.

### 4 Applications

In this section, we shall present some applications. In particular, we introduce in Subsection 4.1 a subclass of stochastic games with a decomposable coarser transition kernel, where the discrete components in the actions and states could directly influence the transition of the random shocks. Games in this subclass are called stochastic games with endogenous shocks.<sup>23</sup> As an illustrative application, we shall consider in Subsection 4.2 a stochastic version of a dynamic oligopoly model as studied in [29, 30].

#### 4.1 Stochastic games with endogenous shocks

Economic models with both discrete and continuous components of actions are common. For example, firms make discrete choices like entry or exit of the market, and continuous choices like quantities and prices of the products. Here we shall consider stochastic games where the discrete choices may play a distinct role. For this purpose, we assume that for each player  $i \in I$ , her action space has two components in the form  $X_i = X_i^d \times X_i^-$ , where  $X_i^d$  is a finite set representing the discrete choices, and  $X_i^-$  is a compact metric space representing possibly other types of actions. Let  $X^d = \prod_{i \in I} X_i^d$  and  $X^- = \prod_{i \in I} X_i^-$ .

 $<sup>^{23}</sup>$ As in [11], we also allow the random shocks to enter into the stage payoffs of the players.

Since actions in the previous period could be part of the current state, the state space may have discrete and continuous components as well. On the other hand, random shocks form a common feature in various economic models. As a result, we assume that the state space has the form  $S = H^d \times H^- \times R$ . Here R models the random shocks. The components  $H^d$  and  $H^-$  represent respectively a finite set of fundamental parameters and the residual part of fundamental parameters in the states.

As defined in Section 2, the state transition in a stochastic game generally depends on the actions and state in the previous period. However, as shown by the counterexample in [28], a stationary Markov perfect equilibrium in a stochastic game may not exist without restriction on the transition kernel. Indeed the restriction, as considered in [10], [11] and [38], is to assume that the transition of random shocks in the stochastic games does not depend on the actions and state in the previous period explicitly. For a stochastic game whose state space is in the form  $S = H^d \times H^- \times R$ , our innovation is to allow that the transition of random shocks directly depends on the discrete components  $H^d$  and  $X^d$  of the state and actions from the previous period.

We shall now describe formally the state space and the law of motion in a stochastic game with endogenous shocks with the rest of parameters and conditions as in Section 2.

- 1. The state space can be written as  $S = H^d \times H^- \times R$  and  $S = \mathcal{H}^d \otimes \mathcal{H}^- \otimes \mathscr{R}$ , where  $H^d$  is a finite set with its power set  $\mathcal{H}^d$ ,  $H^-$  and R are complete separable metric spaces endowed with the Borel  $\sigma$ -algebras  $\mathcal{H}^-$  and  $\mathscr{R}$ respectively. Denote  $H = H^d \times H^-$  and  $\mathcal{H} = \mathcal{H}^d \otimes \mathcal{H}^-$ .
- 2. Recall that  $Q : S \times X \times S \to [0,1]$  is the transition probability. For  $s \in S, x \in X$ , let  $Q_H(\cdot|s, x)$  be the marginal of  $Q(\cdot|s, x)$  on H. There is a fixed probability measure  $\kappa$  on  $(H, \mathcal{H})$  such that for all s and  $x, Q_H(\cdot|s, x)$  is absolutely continuous with respect to  $\kappa$  with the corresponding product measurable Radon-Nikodym derivative  $\phi(\cdot|s, x)$ . For each  $s \in S, Q_H(\cdot|s, x)$  is norm continuous in x.
- 3. The distribution of  $r' \in R$  in the current period depends on  $h' \in H$  in the

current period as well as  $h^d$  and  $x^d$  in the previous period. In particular,  $Q_R: H^d \times X^d \times H \times \mathscr{R} \to [0, 1]$  is a transition probability such that for all  $s = (h^d, h^-, r), x = (x^d, x^-), \text{ and } Z \in \mathcal{S}$ , we have

$$Q(Z|s,x) = \int_{H} \int_{R} \mathbf{1}_{Z}(h',r') Q_{R}(\mathrm{d}r'|h^{d},x^{d},h') Q_{H}(\mathrm{d}h'|s,x).^{24}$$

4. For any  $h^d$ ,  $x^d$  and h',  $Q_R(\cdot|h^d, x^d, h')$  is absolutely continuous with respect to an atomless probability measure  $\nu$  on  $(R, \mathscr{R})$  with the corresponding product measurable Radon-Nikodym derivative  $\psi(\cdot|h^d, x^d, h')$ .

**Remark 1.** A noisy stochastic game as considered in [11] is the case that  $H^d$  and  $X^d$  are singletons. When both  $H^-$  and  $X^-$  are singletons, the random shocks in the current period could fully depend on the action profile and the fundamental part of the state in the previous period since  $Q_R$  could fully depend on  $(h^d, x^d)$  in the previous period; see, for example, the stochastic dynamic oligopoly model in the next subsection. Remark 2 below considers possible applications in which  $H^d$ ,  $X^d$ ,  $H^-$  and  $X^-$  are all non-singletons.

The following result is a simple corollary of Theorem 1.

**Corollary 1.** A stochastic game with endogenous shocks has a decomposable coarser transition kernel, and hence possesses a stationary Markov perfect equilibrium.

*Proof.* Let  $\lambda = \kappa \otimes \nu$ , and  $\mathcal{G} = \mathcal{H} \otimes \{\emptyset, R\}$ . Since  $\nu$  is atomless,  $\mathcal{S}$  has no  $\mathcal{G}$ -atom under  $\lambda$ . It is clear that for all  $s = (h^d, h^-, r), x = (x^d, x^-)$ , and  $Z \in \mathcal{S}$ , we have

$$Q(Z|s,x) = \int_{H \times R} \mathbf{1}_{Z}(h',r')\phi(h'|s,x) \cdot \psi(r'|h^{d},x^{d},h') \,\mathrm{d}\lambda(h',r'),^{25}$$

which means that the corresponding transition kernel

$$q(s'|s, x) = \phi(h'|s, x) \cdot \psi(r'|h^d, x^d, h')$$

<sup>&</sup>lt;sup>24</sup>For any set A, the indicator function  $\mathbf{1}_A$  of A is a function such that  $\mathbf{1}_A(y)$  is 1 if  $y \in A$  and 0 otherwise.

<sup>&</sup>lt;sup>25</sup>For clarity and notational simplicity, we use s, (h, r), and  $(h^d, h^-, r)$  interchangably (similarly for their "prime" versions).

for s' = (h', r').

For any fixed  $\tilde{x}^d \in X^d$  and  $\tilde{h}^d \in H^d$ , define an  $S \otimes S \otimes \mathcal{B}(X)$ -measurable function  $\hat{\phi}_{(\tilde{h}^d, \tilde{x}^d)}$  on  $S \times S \times X$  and an S-measurable function  $\hat{\psi}_{(\tilde{h}^d, \tilde{x}^d)}$  on S such that for any  $s' = (h', r') \in S$ ,  $s = (h^d, h^-, r) \in S$  and  $x = (x^d, x^-) \in X$ ,

- $\hat{\phi}_{\left(\tilde{h}^{d},\tilde{x}^{d}\right)}(s',s,x) = \phi(h'|s,x) \cdot \mathbf{1}_{\left\{\left(\tilde{h}^{d},\tilde{x}^{d}\right)\right\}}\left((h^{d},x^{d})\right);$
- $\hat{\psi}_{\left(\tilde{h}^{d},\tilde{x}^{d}\right)}(s') = \psi(r'|\tilde{h}^{d},\tilde{x}^{d},h').$

It is clear that  $\hat{\phi}_{(\tilde{h}^d, \tilde{x}^d)}(\cdot, s, x)$  is  $\mathcal{G}$ -measurable for any fixed s and x. Then we have

$$q(s'|s,x) = \phi(h'|s,x) \cdot \psi(r'|h',h^d,x^d) = \sum_{\tilde{h}^d \in H^d, \tilde{x}^d \in X^d} \hat{\phi}_{\left(\tilde{h}^d, \tilde{x}^d\right)}(s',s,x) \cdot \hat{\psi}_{\left(\tilde{h}^d, \tilde{x}^d\right)}(s').$$

Since  $H^d$  and  $X^d$  are both finite, the stochastic game with endogenous shocks has a decomposable coarser transition kernel, and a stationary Markov perfect equilibrium by Theorem 1.

# 4.2 Stochastic dynamic oligopoly with price competition

As mentioned in the previous subsection, it is natural to model certain economic situations as stochastic games with endogenous shocks. In this subsection, we shall consider a stochastic analog of a dynamic oligopoly asynchronous choice model as studied in [29, 30]. We verify that such a stochastic dynamic oligopoly model is actually a stochastic game with endogenous shocks. Corollary 1 then allows us to claim the existence of a stationary Markov perfect equilibrium.

Competition between two firms (i = 1, 2) takes place in discrete time with an infinite horizon. Each period begins with a set of firms active in the market, a vector of prices from the previous period, and a demand shock. An active firm will make the price decision while an inactive firm follows its price in the previous period. The firms' profits depend on the prices of both firms and the demand shock. This model is an extension of the model as considered in [30], where firms move alternatively. In contrast, we allow for any possible transition between inactive and active firms, and introduce endogenous demand shocks which is considered desirable in [30, p. 587].

Formally, the player space is  $I = \{1, 2\}$ . An element  $\theta$  in the set  $\Theta = \{\theta_1, \theta_2, \theta_3\}$ indicates which players are active at the beginning of each period. Player 1 (resp. player 2) is active if  $\theta = \theta_1$  (resp.  $\theta_2$ ); both players are active if  $\theta = \theta_3$ . That is, instead of focusing on a fixed short commitment for two periods, we allow for random commitments (see [29] for more discussion). As in the model of [30, p. 573], the price space  $P_1 = P_2 = P$  is assumed to be finite, which means that firms cannot set prices in units smaller than, say, a penny. Let  $\tilde{P} = P$ , and  $\tilde{P}^2 = \tilde{P} \times \tilde{P}$  denote the set of prices from the previous period. Let R be a set of demand shocks, which is a closed subset of the Euclidean space  $\mathbb{R}^l$ . The state of the market is then summarized by a vector  $(\theta, (\tilde{p_1}, \tilde{p_2}), r)$ .

A decision of firm *i* is to propose a price  $a_i \in P_i$ . At state  $(\theta, (\tilde{p_1}, \tilde{p_2}), r)$ , the set of feasible actions for firm *i* is  $P_i$  if  $\theta = \theta_i$  or  $\theta_3$ , and  $\{\tilde{p_i}\}$  otherwise. If a firm is active, then it can pick any price in  $P_i$ . If the firm is inactive, then it must commit to its price from the previous period.

Given the state  $s = (\theta, (\tilde{p_1}, \tilde{p_2}), r)$  and action profile  $(p_1, p_2)$  in the previous period, the state in the current period, denoted by  $(\theta', (\tilde{p_1}', \tilde{p_2}'), r')$ , is determined as follows:

- 1.  $\theta'$  is determined by s and  $(p_1, p_2)$ , following a transition probability  $\kappa_1(\cdot | s, p_1, p_2)$ ;
- 2.  $\tilde{p_i}' = p_i$  for i = 1, 2, which means that the action profile in the previous period is publicly observed and viewed as part of the state in the current period;
- 3. r' is determined by an atomless transition probability  $\mu(\cdot|\theta', p_1, p_2, \theta, \tilde{p_1}, \tilde{p_2})$ , which means that demand shocks are directly determined by the prices and firms' relative positions in the previous and current periods.

Suppose that the market demand function  $D: P \times R \to \mathbb{R}_+$  and the cost function  $c: R \to \mathbb{R}_+$  are both bounded. At the state  $s = (\theta, (\tilde{p_1}, \tilde{p_2}), r)$ , firm *i*'s profit is given by

$$u_i(p_1, p_2, r) = \begin{cases} (p_i - c(r))D(p_i, r) & p_i < p_j; \\ \frac{(p_i - c(r))D(p_i, r)}{2} & p_i = p_j; \\ 0 & p_i > p_j. \end{cases}$$

where  $(p_1, p_2)$  is the action profile in the current period. Firm *i* discounts the future by a discount factor  $0 < \beta_i < 1$ .

We shall show that this stochastic dynamic duopoly model can be viewed as a stochastic game with endogenous shocks. For both firms,  $X_i^d = P_i$  and  $X_i^$ is a set with only one element. Let  $H^d = \Theta \times \tilde{P}^2$ ,  $H^-$  be a singleton set, and  $H = H^d \times H^-$ . Let  $\kappa$  be the counting probability measure on H and

$$\nu = \frac{1}{J} \sum_{(\theta, \theta') \in \Theta^2, (p_1, p_2) \in P_1 \times P_2, (\tilde{p_1}, \tilde{p_2}) \in \tilde{P}^2} \mu(\cdot | \theta', p_1, p_2, \theta, \tilde{p_1}, \tilde{p_2}),$$

where J is cardinality of the product space  $\Theta^2 \times P^4$ . Then the marginal measure  $Q_H(\cdot|s, p_1, p_2)$  is absolutely continuous with respect to  $\kappa$ , and  $Q_R(\cdot|h', h^d, p_1, p_2) = \mu(\cdot|\theta', p_1, p_2, \theta, \tilde{p_1}, \tilde{p_2})$  is absolutely continuous with respect to  $\nu$ . Since the space of action profiles is finite, the continuity requirement on the payoff functions and transition kernel is automatically satisfied. Therefore, the following proposition follows from Corollary 1.

**Proposition 1.** The above dynamic duopoly model has a stationary Markov perfect equilibrium.

**Remark 2.** In the above model, the position of a firm (active or inactive) is randomly determined, and hence is not a choice variable. One can consider a dynamic market model in which the positions of the firms are determined by the endogenous entry/exit decisions; see, for example, [11], [13] and [20]. In particular, in the application on firm entry, exit, and investment in [11], each firm needs to make decisions on entry/exit as well as on the production plan. Thus, the action space naturally has two components. The state space can be divided as a product of three parts: the first part Z provides a list of firms which are present or absent in the market in the current period; the second part K is the available capital stock in the current period; and the last part R represents exogenously given random shocks which are i.i.d with a density with respect to the Lebesgue measure. This was formulated in [11] as a noisy stochastic game, which can also be viewed as a stochastic game with endogenous shocks with  $H^d$  as a singleton set and  $H^- = Z \times$ K. This model can be extended by letting  $H^d = Z$ ,  $H^- = K$ , and the transition of R depend on firms' positions in the previous and current periods. Such dependence of the random shocks allows for the interpretation that the change in the number of active firms as well as the shift of market positions by some firms<sup>26</sup> do matter for the demand/supply shocks. For example, the decision of a big firm to quit the market is a shock that could significantly distort the expectation of the demand side, while the exit decision of a small firm may not even be noticed. The extended model remains a stochastic game with endogenous shocks, which has a stationary Markov perfect equilibrium.<sup>27</sup>

### 5 An Extension

In Section 3, we assume that the probability measure  $\lambda$  is atomless on (S, S). Below, we consider the more general case that  $\lambda$  may have atoms. To guarantee the existence of stationary Markov perfect equilibria, we still assume the condition of decomposable coarser transition kernel, but only on the atomless part.

- 1. There exist disjoint S-measurable subsets  $S_a$  and  $S_l$  such that  $S_a \cup S_l = S$ ,  $\lambda|_{S_a}$  is the atomless part of  $\lambda$  while  $\lambda|_{S_l}$  is the purely atomic part of  $\lambda$ . The subset  $S_l$  is countable and each singleton  $\{s_l\}$  with  $s_l \in S_l$  is S-measurable with  $\lambda(s_l) > 0$ .<sup>28</sup>
- 2. For  $s_a \in S_a$ , the transition kernel  $q(s_a|s,x) = \sum_{1 \le j \le J} q_j(s_a,s,x) \rho_j(s_a)$  for

 $<sup>^{26}</sup>$ If 1 and 0 represent active or inactive firms respectively, then a shift of market position for the active (inactive) firm means to change its position from 1 to 0 (0 to 1).

<sup>&</sup>lt;sup>27</sup>As noted in the introduction, our existence results on stochastic games, whose state transition for the shock component explicitly depends on the parameters in the previous stage, cannot be covered by earlier results. These include Corollary 1, Proposition 1 as well as the results suggested in Remark 2.

<sup>&</sup>lt;sup>28</sup>This assumption is only for simplicity. It is immediate to extend our result to the case that  $S_l$  is a collection of at most countably many atoms in the usual measure-theoretic sense. Note that for a set  $\{s\}$  with one element in S, we use  $\lambda(s)$  to denote the measure of this set instead of  $\lambda(\{s\})$ .

some positive integer J, and for each  $s \in S$  and  $x \in X$ , where  $q_j$  is product measurable, and  $q_j(\cdot, s, x)$  and  $\rho_j$  are nonnegative and integrable on the atomless measure space  $(S_a, \mathcal{S}^{S_a}, \lambda|_{S_a})$  for  $j = 1, \ldots, J$ .<sup>29</sup>

**Definition 2.** Let  $\mathcal{G}$  be a sub- $\sigma$ -algebra of  $\mathcal{S}^{S_a}$ . A discounted stochastic game is said to have a **decomposable coarser transition kernel on the atomless part** if  $\mathcal{S}^{S_a}$  has no  $\mathcal{G}$ -atom under  $\lambda|_{S_a}$  and  $q_j(\cdot, s, x)$  is  $\mathcal{G}$ -measurable on  $S_a$  for each  $s \in S$  and  $x \in X$ ,  $j = 1, \ldots, J$ .

The following theorem shows that the equilibrium existence result still holds.

**Theorem 2.** Every discounted stochastic game with a decomposable coarser transition kernel on the atomless part has a stationary Markov perfect equilibrium.

### 6 Discussion

In this section, we shall discuss the relationship between our results and several related results. In particular, we show that our results cover the existence results on correlated equilibria, noisy stochastic games, stochastic games with state-independent transitions, and stochastic games with mixtures of constant transition kernels as special cases. We also explicitly demonstrate why a recent counterexample fails to satisfy our conditions, and discuss the minimality of our conditions.

#### Correlated equilibria

It was proved in [38] that a correlated stationary Markov perfect equilibrium exists in discounted stochastic games in the setup described in our Section 2. Ergodic properties of such correlated equilibria were obtained in [10] under stronger conditions. They essentially assumed that players can observe the outcome of a public randomization device before making decisions at each stage.<sup>30</sup> Thus, the new state space can be regarded as  $S' = S \times L$  endowed with the product

<sup>&</sup>lt;sup>29</sup>It is clear that for any  $E \in S$ , the transition probability  $Q(E|s,x) = \int_{E \cap S_a} q(s_a|s,x) d\lambda(s_a) + \sum_{\substack{s_l \in S_l \\ 30}} \mathbf{1}_E(s_l) q(s_l|s,x) \lambda(s_l)$  for any  $s \in S$  and  $x \in X$ .

<sup>&</sup>lt;sup>30</sup>For detailed discussions on such a public randomization device, or "sunspot", see [10] and their references.

 $\sigma$ -algebra  $S' = S \otimes B$  and product measure  $\lambda' = \lambda \otimes \eta$ , where L is the unit interval endowed with the Borel  $\sigma$ -algebra  $\mathcal{B}$  and Lebesgue measure  $\eta$ . Denote  $\mathcal{G}' = S \otimes \{\emptyset, L\}$ . Given  $s', s'_1 \in S'$  and  $x \in X$ , the new transition kernel  $q'(s'_1|s', x) =$  $q(s_1|s, x)$ , where s (resp.  $s_1$ ) is the projection of s' (resp.  $s'_1$ ) on S and q is the original transition kernel with the state space S. Thus,  $q'(\cdot|s', x)$  is measurable with respect to  $\mathcal{G}'$  for any  $s' \in S'$  and  $x \in X$ . It is obvious that S' has no  $\mathcal{G}'$ -atom. Then the condition of coarser transition kernel is satisfied for the extended state space  $(S', S', \lambda')$ , and the existence of a stationary Markov perfect equilibrium follows from Theorem 1.<sup>31</sup> The drawback of this approach is that the "sunspot" is irrelevant to the fundamental parameters of the game. Our result shows that it can indeed enter the stage payoff u, the correspondence of feasible actions A and the transition probability Q.

### Stochastic games with finite actions and state-independent transitions

In [40], they studied stochastic games with finite actions and state-independent atomless transitions. Namely,  $X_i$  is finite for each  $i \in I$  and the transition probability Q does not directly depend on s (to be denoted by  $Q(\cdot|x)$ ). We shall check that such a stochastic game satisfies the condition of decomposable coarser transition kernels.

Let  $\lambda = \sum_{x \in X} Q(\cdot|x)$ . Then, for each x,  $Q(\cdot|x)$  is absolutely continuous with respect to  $\lambda$ , and the corresponding Radon-Nikodym derivative is denoted by  $q(\cdot|x)$ (abbreviated as  $q_x$ ). Let  $\delta_x$  be the Dirac function on X:  $\delta_x(y) = 1$  if y = x, and 0 otherwise. Then we have

$$q(s'|x) = \sum_{y \in X} \delta_y(x) q_y(s').$$

It is obvious that the condition of decomposable coarser transition kernels is satisfied with  $\mathcal{G} = \{\emptyset, S\}$ . Then a stationary Markov perfect equilibrium exists by Theorem 1.

 $<sup>^{31}</sup>$ As noted in [38], a stochastic version of Caratheodory's theorem implies that one can find a stationary Markov correlated equilibrium strategy as a stochastic convex combination of stationary Markov strategies.

#### Decomposable constant transition kernels on the atomless part

Stochastic games with transition probabilities as combinations of finitely many measures on the atomless part were considered in [35]. In particular, the structure of the transition probability in [35] is as follows.

- 1.  $S_2$  is a countable subset of S and  $S_1 = S \setminus S_2$ , each point in  $S_2$  is S-measurable.
- 2. There are atomless nonnegative measures  $\mu_j$  concentrated on  $S_1$ , nonnegative measures  $\delta_k$  concentrated on  $S_2$ , and measurable functions  $q_j, b_k : S \times X \to [0,1], 1 \leq j \leq J$  and  $1 \leq k \leq K$ , where J and K are positive integers. The transition probability  $Q(\cdot|s,x) = \delta(\cdot|s,x) + Q'(\cdot|s,x)$  for each  $s \in S$  and  $x \in X$ , where  $\delta(\cdot|s,x) = \sum_{1 \leq k \leq K} b_k(s,x)\delta_k(\cdot)$  and  $Q'(\cdot|s,x) = \sum_{1 \leq j \leq J} q_j(s,x)\mu_j(\cdot)$ .
- 3. For any j and k,  $q_j(s, \cdot)$  and  $b_k(s, \cdot)$  are continuous on X for any  $s \in S$ .

We shall show that any stochastic game with the above structure satisfies the condition of decomposable coarser transition kernel on the atomless part.

Without loss of generality, assume that  $\mu_j$  and  $\delta_k$  are all probability measures. Let  $\lambda(E) = \frac{1}{J+K} \left( \sum_{1 \leq j \leq J} \mu_j(E) + \sum_{1 \leq k \leq K} \delta_k(E) \right)$  for any  $E \in \mathcal{S}$ . Then  $\mu_j$  is absolutely continuous with respect to  $\lambda$  and assume that  $\rho_j$  is the Radon-Nikodym derivative for  $1 \leq j \leq J$ .

Given any  $s \in S$  and  $x \in X$ , let

$$q(s'|s, x) = \begin{cases} \sum_{1 \le j \le J} q_j(s, x) \rho_j(s'), & \text{if } s' \in S_1; \\ \frac{\delta(s'|s, x)}{\lambda(s')}, & \text{if } s' \in S_2 \text{ and } \lambda(s') > 0; \\ 0, & \text{if } s' \in S_2 \text{ and } \lambda(s') = 0. \end{cases}$$

Then  $Q(\cdot|s, x)$  is absolutely continuous with respect to  $\lambda$  and  $q(\cdot|s, x)$  is the transition kernel. The condition of a decomposable coarser transition kernel on the atomless part is satisfied with  $\mathcal{G} = \{\emptyset, S_1\}$ . Then a stationary Markov perfect equilibrium exists by Theorem 2.

#### Noisy stochastic games

It was proved in [11] that stationary Markov perfect equilibria exist in stochastic games with noise – a component of the state that is nonatomically distributed and not directly affected by the previous period's state and actions. As indicated in Remark 1, a noisy stochastic game is a special case of a stochastic game with endogenous shocks such that  $H^d$  and  $X^d$  are both singletons.

By Corollary 1, a noisy stochastic games has a decomposable coarser transition kernel. Below, we directly show that any noisy stochastic game indeed has a coarser transition kernel. The existence of stationary Markov perfect equilibria in noisy stochastic games thus follows from Theorem 1. The proof of the proposition below is left in the appendix.

**Proposition 2.** Every noisy stochastic game has a coarser transition kernel.

#### Nonexistence of stationary Markov perfect equilibrium

A concrete example of a stochastic game satisfying all the conditions as stated in Section 2 was presented in [28], which has no stationary Markov perfect equilibrium. Their example will be described in the following.

- 1. The player space is  $\{A, B, C, C', D, D', E, F\}$ .
- Player A has the action space {U, D}, and player B has the action space {L, M, R}. Players C, C', D, D' have the same action space {0,1}. Players E, F have the action space {-1,1}.
- 3. The state space is S = [0, 1] endowed with the Borel  $\sigma$ -algebra  $\mathcal{B}$ .
- 4. For any action profile x, let

$$h(x) = x_C + x_{C'} + x_D + x_{D'},$$

where  $x_i$  is the action of player *i*. For each  $s \in [0, 1]$ , let

$$\tilde{Q}(s,x) = (1-s) \cdot \frac{1}{64} h(x),$$

and U(s, 1) be the uniform distribution on [s, 1] for  $s \in [0, 1)$ . The transition

probability is given by

$$Q(s,x) = \tilde{Q}(s,x)U(s,1) + (1 - \tilde{Q}(s,x))\delta_1,$$

where  $\delta_1$  is the Dirac measure at 1.

The following proposition shows that the condition of a decomposable coarser transition kernel on the atomless part is violated in this example.<sup>32</sup>

**Proposition 3.** The atomless part  $\tilde{Q}(s, x)U(s, 1)$  of the transition probability in the Example of [28] does not have a decomposable coarser transition kernel.

#### Minimality

We have shown the existence of stationary Markov perfect equilibria in discounted stochastic games by assuming the presence of a (decomposable) coarser transition kernel. This raises the question of whether our condition is minimal and, if so, then in what sense.

As discussed in the introduction, the main difficulty in the existence argument for stochastic games is due to the failure of the convexity of the equilibrium payoff correspondence P of a one-shot auxiliary game as parameterized by state s and continuation value function v. As will be shown in Subsection 8.2, the correspondence R(v), which is the collection of selections from the equilibrium payoff correspondence  $P(v, \cdot)$ , will live in an infinite-dimensional space if there is a continuum of states. Thus, the desirable closedness and upper hemicontinuity properties would fail even though P has these properties in terms of v. To handle such issues, the standard approach is to work with the convex hull co(R). We bypass this imposed convexity restriction by using the result that  $\mathcal{I}_G^{(S,\mathcal{G},\lambda)} = \mathcal{I}_{co(G)}^{(S,\mathcal{G},\lambda)}$ for any S-measurable, integrably bounded,<sup>33</sup> closed valued correspondence Gprovided that S has no  $\mathcal{G}$ -atom. Moreover, for the condition of a decomposable coarser transition kernel, we assume that the transition kernel can be divided into

 $<sup>^{32}</sup>$ Proposition 3 can also be implied by the nonexistence result in [28] and our Theorem 2. However, the argument in [28] is deep, and our proof explicitly demonstrates why their example fails to satisfy our sufficient condition in Theorem 2.

<sup>&</sup>lt;sup>33</sup>A correspondence  $G: (\mathcal{S}, \mathcal{S}, \lambda) \to \mathbb{R}^n$  is said to be integrably bounded if there exists some integrable function  $\varphi: (\mathcal{S}, \mathcal{S}, \lambda) \to \mathbb{R}_+$  such that  $||G(s)|| \le \varphi(s)$  for  $\lambda$ -almost all s, where  $|| \cdot ||$  is usual norm on  $\mathbb{R}^n$ .

finitely many parts. The following propositions demonstrate the minimality of our condition from a technical point of view.

**Proposition 4.** Suppose that  $(S, S, \lambda)$  has a  $\mathcal{G}$ -atom D with  $\lambda(D) > 0$ . Then there exists a measurable correspondence G from  $(S, S, \lambda)$  to  $\{0, 1\}$  such that  $\mathcal{I}_G^{(S, \mathcal{G}, \lambda)} \neq \mathcal{I}_{CO(G)}^{(S, \mathcal{G}, \lambda)}$ .

The key result that we need in the proof of Theorem 1 is  $\mathcal{I}_{H}^{(\mathcal{S},\mathcal{G},\lambda)} = \mathcal{I}_{\mathrm{CO}(H)}^{(\mathcal{S},\mathcal{G},\lambda)}$ . The question is whether a similar result holds if we generalize the condition of a decomposable coarser transition kernel from a finite sum to a countable sum. We will show that this is not possible.

Let  $(S, S, \lambda)$  be the Lebesgue unit interval  $(L, \mathcal{B}, \eta)$ . Suppose that  $\{\varrho_n\}_{n \ge 0}$  is a complete orthogonal system in  $L_2(S, S, \lambda)$  such that  $\varrho_n$  takes value in  $\{-1, 1\}$  and  $\int_S \varrho_n d\lambda = 0$  for each  $n \ge 1$  and  $\varrho_0 \equiv 1$ . Let  $\rho_n = \varrho_n + 1$  for each  $n \ge 1$  and  $\rho_0 = \varrho_0$ . Let  $\{E_n\}_{n\ge 0}$  be a countable measurable partition of S and  $q_n(s) = \mathbf{1}_{E_n}$  for each  $n \ge 0$ . Suppose that a transition kernel q is decomposed into a countable sum  $q(s_1|s,x) = \sum_{n\ge 0} q_n(s)\rho_n(s_1)$ . The following proposition shows that the argument for the case that J is finite is not valid for such an extension.<sup>34</sup>

**Proposition 5.** Let  $G(s) = \{-1, 1\}$  for  $s \in S$ , and f be the measurable selection of co(G) that takes the constant value 0. Then, for any sub- $\sigma$ -algebra  $\mathcal{F} \subseteq S$ , there is no S-measurable selection g of G with  $\mathbb{E}^{\lambda}(g\rho_n|\mathcal{F}) = \mathbb{E}^{\lambda}(f\rho_n|\mathcal{F})$  for any  $n \geq 0$ .

Thus, our condition is minimal in the sense that if one would like to adopt the "one-shot game" approach as used in the literature for obtaining a stationary Markov perfect equilibrium, then it is a tight condition.

### 7 Concluding Remarks

We consider stationary Markov perfect equilibria in discounted stochastic games with a general state space. So far, such equilibria have been shown to exist only for several special classes of stochastic games. In this paper, the existence of stationary Markov perfect equilibria is proved under some general conditions, which

<sup>&</sup>lt;sup>34</sup>It is a variant of a well known example of Lyapunov.

broadens the scope of potential applications of stochastic games. We illustrate such applications via two examples, namely, stochastic games with endogenous shocks and a stochastic dynamic oligopoly model. Our results unify and go beyond various existence results as in [10], [11], [35], [38] and [40], and also provide some explanation for a recent counterexample which fails to have an equilibrium in stationary strategies.

In the literature, the standard recursive approach for the existence arguments is to work with a one-shot auxiliary game parameterized by the state and the continuation value function. We adopt this approach and provide a very simple proof under the condition of "(decomposable) coarser transition kernels". The proof is based on a new connection between stochastic games and conditional expectations of correspondences. We demonstrate from a technical point of view that our condition is minimal for the standard "one-shot game" approach.

# 8 Appendix

### 8.1 Proof of Claim 1

Let  $\mathcal{F}$  be the sub- $\sigma$ -algebra of  $\mathcal{S}$  that is generated by the collection of functions  $\{q(\cdot,s)\}_{s\in S}$ . Let  $(\hat{h},\hat{r}) = (0,0)$  (resp.  $(\tilde{h},\tilde{r}) = (\frac{1}{3},0)$ ), and denote  $\hat{q}$  (resp.  $\tilde{q}$ ) as the corresponding value of  $q(s_1,\hat{h},\hat{r})$  (resp.  $q(s_1,\tilde{h},\tilde{r})$ ). Then both  $\hat{q}$  and  $\tilde{q}$  are functions of  $s_1$  and  $\mathcal{F}$ -measurable. We have the following system of equations:

$$\hat{q} = \frac{3}{2}h_1^2 + 2h_1r_1, \tag{3}$$

$$\tilde{q} = h_1^2 + \frac{1}{3}h_1 + 2h_1r_1.$$
(4)

We can view  $h_1$  and  $r_1$  as two functions of  $(\hat{q}, \tilde{q})$ . By subtracting Equation (4) from Equation (3), we get  $\hat{q} - \tilde{q} = \frac{1}{2}h_1^2 - \frac{1}{3}h_1$ . By the definition, this equation must have solutions in terms of  $h_1$  for the given value  $\hat{q} - \tilde{q}$ . Thus, we have  $\frac{1}{9} + 2(\hat{q} - \tilde{q}) \ge 0$ , and  $h_1 = \frac{1}{3} + \sqrt{\frac{1}{9} + 2(\hat{q} - \tilde{q})}$  or  $\frac{1}{3} - \sqrt{\frac{1}{9} + 2(\hat{q} - \tilde{q})}$ , with at least one of them in [0,1]. We can denote

$$\hat{\alpha}(\hat{q}, \tilde{q}) = \begin{cases} \frac{1}{3} + \sqrt{\frac{1}{9} + 2(\hat{q} - \tilde{q})}, & \text{if } \frac{1}{3} + \sqrt{\frac{1}{9} + 2(\hat{q} - \tilde{q})} \in [0, 1]; \\ \frac{1}{3} - \sqrt{\frac{1}{9} + 2(\hat{q} - \tilde{q})}, & \text{otherwise.} \end{cases}$$

By substituting  $h_1 = \hat{\alpha}(\hat{q}, \tilde{q})$  into Equation (3), we can solve  $r_1$  as a function of  $(\hat{q}, \tilde{q})$ , which is denoted by  $r_1 = \tilde{\alpha}(\hat{q}, \tilde{q})$ .<sup>35</sup>

Let  $\pi_h$  and  $\pi_r$  be the projection mappings on  $S_1$  with  $\pi_h(h_1, r_1) = h_1$  and  $\pi_h(h_1, r_1) = r_1$  respectively. Because both  $\hat{q}$  and  $\tilde{q}$  are  $\mathcal{F}$ -measurable functions on  $S_1$ , and  $\pi_h = \hat{\alpha}(\hat{q}, \tilde{q})$  and  $\pi_r = \tilde{\alpha}(\hat{q}, \tilde{q})$ , the mappings  $\pi_h$  and  $\pi_r$  are also  $\mathcal{F}$ -measurable. As a result, both  $\mathcal{B}([0, 1]) \otimes \{\emptyset, [0, 1]\}$  and  $\{\emptyset, [0, 1]\} \otimes \mathcal{B}([0, 1])$  are contained in  $\mathcal{F}$ . Thus,  $\mathcal{S} \subseteq \mathcal{F}$ . Since  $\mathcal{F} \subseteq \mathcal{S}$  by the definition of  $\mathcal{F}$ , we have  $\mathcal{F} = \mathcal{S}$ .

### 8.2 Proof of Theorem 1

In this subsection, we shall prove Theorem 1.

Let  $L_1((S, \mathcal{S}, \lambda), \mathbb{R}^m)$  and  $L_{\infty}((S, \mathcal{S}, \lambda), \mathbb{R}^m)$  be the  $L_1$  and  $L_{\infty}$  spaces of all  $\mathcal{S}$ -measurable mappings from S to  $\mathbb{R}^m$  with the usual norm; that is,

$$L_1((S, \mathcal{S}, \lambda), \mathbb{R}^m) = \{ f \colon f \text{ is } \mathcal{S}\text{-measurable and } \int_S \|f\| \, \mathrm{d}\lambda < \infty \},$$

 $L_{\infty}((S, \mathcal{S}, \lambda), \mathbb{R}^m) = \{ f \colon f \text{ is } \mathcal{S}\text{-measurable and essentially bounded under } \lambda \},\$ 

where  $\|\cdot\|$  is the usual norm in  $\mathbb{R}^m$ . By the Riesz representation theorem (see Theorem 13.28 of [1]),  $L_{\infty}((S, S, \lambda), \mathbb{R}^m)$  can be viewed as the dual space of  $L_1((S, S, \lambda), \mathbb{R}^m)$ . Then  $L_{\infty}((S, S, \lambda), \mathbb{R}^m)$  is a locally convex, Hausdorff topological vector space under the weak\* topology. Let  $V = \{v \in$  $L_{\infty}((S, S, \lambda), \mathbb{R}^m)$ :  $\|v\|_{\infty} \leq C\}$ , where C is the upper bound of the stage payoff function u and  $\|\cdot\|_{\infty}$  is the essential sup norm of  $L_{\infty}((S, S, \lambda), \mathbb{R}^m)$ . Then V is nonempty and convex. Moreover, V is compact under the weak\* topology by Alaoglu's Theorem (see Theorem 6.21 of [1]). Since S is countably generated,  $L_1((S, S, \lambda), \mathbb{R}^m)$  is separable, which implies that V is metrizable in the weak\*

<sup>&</sup>lt;sup>35</sup>By the construction, both  $\hat{\alpha}$  and  $\tilde{\alpha}$  are obviously Borel measurable functions.

topology (see Theorem 6.30 of [1]).

Given any  $v = (v_1, \dots, v_m) \in V$  and  $s \in S$ , we consider the game  $\Gamma(v, s)$ . The action space for player *i* is  $A_i(s)$ . The payoff of player *i* with the action profile  $x \in A(s)$  is given by

$$U_i(s,x)(v) = (1 - \beta_i)u_i(s,x) + \beta_i \int_S v_i(s_1)Q(\mathrm{d}s_1|s,x).$$
(5)

A mixed strategy of player *i* is an element in  $\mathcal{M}(A_i(s))$ , and a mixed strategy profile is an element in  $\bigotimes_{i \in I} \mathcal{M}(A_i(s))$ . The set of mixed strategy Nash equilibria of the static game  $\Gamma(v, s)$ , denoted by N(v, s), is a nonempty compact subset of  $\bigotimes_{i \in I} \mathcal{M}(X_i)$  under the weak<sup>\*</sup> topology due to the Fan-Glicksberg Theorem (see [16] and Corollary 17.55 in [1]). Let P(v, s) be the set of payoff vectors induced by the Nash equilibria in N(v, s), and  $\operatorname{co}(P)$  the convex hull of P. Then  $\operatorname{co}(P)$ is a correspondence from  $V \times S$  to  $\mathbb{R}^m$ . Let R(v) (resp.  $\operatorname{co}(R(v))$ ) be the set of  $\lambda$ -equivalence classes of S-measurable selections of  $P(v, \cdot)$  (resp.  $\operatorname{co}(P(v, \cdot))$ ) for each  $v \in V$ .

Following the arguments in Lemmas 6 and 7 in [38] (see also [32]),<sup>36</sup> for each  $v \in V$ ,  $P(v, \cdot)$  (abbreviated as  $P_v(\cdot)$ ) is S-measurable and compact valued, and co(R(v)) is nonempty, convex, weak<sup>\*</sup> compact valued and upper hemicontinuous. Then the correspondence  $co(R) : V \to V$  maps the nonempty, convex, weak<sup>\*</sup> compact set V (a subset of a locally convex Hausdorff topological vector space) to nonempty, convex subsets of V, and it has a closed graph in the weak<sup>\*</sup> topology. By the classical Fan-Glicksberg Fixed Point Theorem, there is a fixed point  $v' \in V$  such that  $v' \in co(R)(v')$ . That is, v' is an S-measurable selection of  $co(P(v', \cdot))$ .

Recall that a correspondence  $G: S \to \mathbb{R}^n$  is said to be integrably bounded if there exists some integrable function  $\varphi$  such that  $||G(s)|| \leq \varphi(s)$  for  $\lambda$ -almost all s, where  $|| \cdot ||$  is usual norm on  $\mathbb{R}^n$ . For any integrably bounded correspondence Gfrom S to  $\mathbb{R}^m$ , define

 $\mathcal{I}_{G}^{(\mathcal{S},\mathcal{G},\lambda)} = \{ \mathbb{E}^{\lambda}(g|\mathcal{G}) \colon g \text{ is an } \mathcal{S}\text{-measurable selection of } G \},\$ 

<sup>&</sup>lt;sup>36</sup>In [38], a slightly stronger condition was imposed on the transition probability Q that the mapping  $q(\cdot|s, x)$  satisfies the  $L_1$  continuity condition in x for all  $s \in S$ . Their arguments on the convexity, compactness and upper hemicontinuity properties still hold in our setting.

where the conditional expectation is taken with respect to  $\lambda$ .

The following lemma is from [12, Theorem 1.2].

**Lemma 1.** If S has no  $\mathcal{G}$ -atom,<sup>37</sup> then for any S-measurable,<sup>38</sup>  $\lambda$ -integrably bounded, closed valued correspondence G,  $\mathcal{I}_G^{(S,\mathcal{G},\lambda)} = \mathcal{I}_{CO(G)}^{(S,\mathcal{G},\lambda)}$ .

Now we are ready to prove Theorem 1.

*Proof.* Given v', let

$$H(s) = \{(a, a \cdot \rho_1(s), \dots, a \cdot \rho_J(s)) : a \in P_{v'}(s)\},\$$

and  $\operatorname{co}(H(s))$  the convex hull of H(s) for each  $s \in S$ . It is clear that H is  $\mathcal{S}$ measurable,  $\lambda$ -integrably bounded and closed valued. Then  $\mathcal{I}_{H}^{(\mathcal{S},\mathcal{G},\lambda)} = \mathcal{I}_{\operatorname{co}(H)}^{(\mathcal{S},\mathcal{G},\lambda)}$  by Lemma 1, which implies that there exists an  $\mathcal{S}$ -measurable selection  $v^*$  of  $P_{v'}$  such that  $\mathbb{E}^{\lambda}(v^*\rho_j|\mathcal{G}) = \mathbb{E}^{\lambda}(v'\rho_j|\mathcal{G})$  for each  $1 \leq j \leq J$ . For each  $i \in I$ ,  $s \in S$  and  $x \in X$ , we have

$$\begin{split} \int_{S} v_i^*(s_1) Q(\mathrm{d}s_1 | s, x) &= \sum_{1 \leq j \leq J} \int_{S} v_i^*(s_1) q_j(s_1, s, x) \rho_j(s_1) \lambda(\mathrm{d}s_1) \\ &= \sum_{1 \leq j \leq J} \int_{S} \mathbb{E}^{\lambda} (v_i^* \rho_j q_j(\cdot, s, x) | \mathcal{G})(s_1) \lambda(\mathrm{d}s_1) \\ &= \sum_{1 \leq j \leq J} \int_{S} \mathbb{E}^{\lambda} (v_i^* \rho_j | \mathcal{G})(s_1) q_j(s_1, s, x) \lambda(\mathrm{d}s_1) \\ &= \sum_{1 \leq j \leq J} \int_{S} \mathbb{E}^{\lambda} (v_i' \rho_j q_j(\cdot, s, x) | \mathcal{G})(s_1) \lambda(\mathrm{d}s_1) \\ &= \sum_{1 \leq j \leq J} \int_{S} \mathbb{E}^{\lambda} (v_i' \rho_j q_j(\cdot, s, x) | \mathcal{G})(s_1) \lambda(\mathrm{d}s_1) \\ &= \sum_{1 \leq j \leq J} \int_{S} v_i'(s_1) q_j(s_1, s, x) \rho_j(s_1) \lambda(\mathrm{d}s_1) \\ &= \int_{S} v_i'(s_1) Q(\mathrm{d}s_1 | s, x). \end{split}$$

<sup>37</sup>In [12], a set  $D \in S$  is said to be a  $\mathcal{G}$ -atom if  $\lambda(D) > 0$  and given any  $D_0 \in S^D$ ,  $\lambda(s \in S : 0 < \lambda(D_0 | \mathcal{G})(s) < \lambda(D | \mathcal{G})(s)) = 0$ . The conditions that  $\mathcal{S}$  has no  $\mathcal{G}$ -atom as in [12] as well as in our paper and the book [26] are equivalent; see Lemma 2 in [18].

<sup>&</sup>lt;sup>38</sup>In [12], the correspondence G is said to be measurable if for any  $x \in \mathbb{R}^m$ , the function d(x, G(s)) is measurable, where d is he usual metric in the Euclidean space. Their notion of measurability of a correspondence coincides with our definition of the measurability of a correspondence, see Theorem 18.5 in [1].

By Equation (5),  $\Gamma(v^*, s) = \Gamma(v', s)$  for any  $s \in S$ , and hence  $P(v^*, s) = P(v', s)$ . Thus,  $v^*$  is an *S*-measurable selection of  $P_{v^*}$ .

By the definition of  $P_{v^*}$ , these exists an S-measurable mapping  $f^*$  from S to  $\bigotimes_{i \in I} \mathcal{M}(X_i)$  such that  $f^*(s)$  is a mixed strategy Nash equilibrium of the game  $\Gamma(v^*, s)$  and  $v^*(s)$  is the corresponding equilibrium payoff for each  $s \in S$ .<sup>39</sup> It is clear that Equations (1) and (2) hold for  $v^*$  and  $f^*$ , which implies that  $f^*$  is a stationary Markov perfect equilibrium.

### 8.3 Proof of Theorem 2

Let  $V_a$  be the set of  $\lambda$ -equivalence classes of S-measurable mappings from  $S_a$  to  $\mathbb{R}^m$  bounded by C. For each  $i \in I$ , let  $F_i$  be the set of all  $f_i \colon S_l \to \mathcal{M}(X_i)$  such that  $f_i(s)(A_i(s)) = 1$  for all  $s \in S_l$ ,  $F = \prod_{i \in I} F_i$ . Let  $V_l$  be the set of mappings from  $S_l$  to  $\mathbb{R}^m$  bounded by C, which is endowed with the supremum metric and hence a complete metric space.

Given  $s \in S$ ,  $v^a \in V_a$  and  $v^l \in V_l$ , consider the game  $\Gamma(v^a, v^l, s)$ . The action space for player *i* is  $A_i(s)$ . The payoff of player *i* with the action profile  $x \in A(s)$ is given by

$$\Phi_{i}(s, x, v^{a}, v^{l}) = (1 - \beta_{i})u_{i}(s, x) + \beta_{i} \sum_{1 \le j \le J} \int_{S_{a}} v_{i}^{a}(s_{a})q_{j}(s_{a}|s, x)\rho_{j}(s_{a})\lambda(\mathrm{d}s_{a}) + \beta_{i} \sum_{s_{l} \in S_{l}} v_{i}^{l}(s_{l})q(s_{l}|s, x)\lambda(s_{l}).$$
(6)

The set of mixed-strategy Nash equilibria in the game  $\Gamma(v^a, v^l, s)$  is denoted as  $N(v^a, v^l, s)$ . Let  $P(v^a, v^l, s)$  be the set of payoff vectors induced by the Nash equilibria in  $N(v^a, v^l, s)$ , and co(P) the convex hull of P.

Given  $v^a \in V_a$ ,  $f \in F$ , define a mapping  $\Pi$  from  $V_l$  to  $V_l$  such that for each  $i \in I, v^l \in V_l$  and  $s_l \in S_l$ ,

$$\Pi_{i}(v^{a}, f_{-i})(v^{l})(s_{l}) = \max_{\phi_{i} \in F_{i}} \int_{X_{-i}} \int_{X_{i}} \Phi_{i}(s_{l}, x_{i}, x_{-i}, v^{a}, v^{l})\phi_{i}(\mathrm{d}x_{i}|s_{l})f_{-i}(\mathrm{d}x_{-i}|s_{l}).$$
(7)

<sup>&</sup>lt;sup>39</sup>Note that  $v^*$  is indeed the corresponding equilibrium payoff for  $\lambda$ -almost all  $s \in S$ . However, one can modify  $v^*$  on a null set such that the claim holds for all  $s \in S$ ; see, for example, [38].

Let  $\beta = \max\{\beta_i \colon i \in I\}$ . Then for any  $v^a \in V_a, v^l, \bar{v}^l \in V_l, x \in X$  and  $s \in S_l$ ,

$$\begin{split} \left| \Phi_i(s, x, v^a, v^l) - \Phi_i(s, x, v^a, \bar{v}^l) \right| &\leq \beta_i \sum_{s_l \in S_l} \left| v_i^l(s_l) - \bar{v}_i^l(s_l) \right| q(s_l | s, x) \lambda(s_l) \\ &\leq \beta_i \sup_{s_l \in S_l} \left| v_i^l(s_l) - \bar{v}_i^l(s_l) \right| \leq \beta \sup_{s_l \in S_l} \left| v_i^l(s_l) - \bar{v}_i^l(s_l) \right|. \end{split}$$

Thus,  $\Pi$  is a  $\beta$ -contraction mapping. There is a unique  $\bar{v}^l \in V_l$  such that  $\Pi_i(v^a, f_{-i})(\bar{v}^l)(s_l) = \bar{v}^l_i(s_l)$  for each  $i \in I$  and  $s_l \in S_l$ . Let  $W(v^a, f)$  be the set of all  $\phi \in F$  such that for each  $i \in I$  and  $s_l \in S_l$ ,

$$\bar{v}_i^l(s_l) = \int_{X_{-i}} \int_{X_i} \Phi_i(s_l, x_i, x_{-i}, v^a, \bar{v}^l) \phi_i(\mathrm{d}x_i | s_l) f_{-i}(\mathrm{d}x_{-i} | s_l).$$
(8)

Let  $\bar{v}^l$  be the function on  $S_l$  generated by  $v^a$  and f as above, and  $R(v^a, f)$ the set of  $\lambda$ -equivalence classes of S-measurable selections of  $P(v^a, \bar{v}^l, \cdot)$  restricted to  $S_a$ . Then, the convex hull  $co(R(v^a, f))$  is the set of  $\lambda$ -equivalence classes of S-measurable selections of  $co(P(v^a, \bar{v}^l, \cdot))$  restricted to  $S_a$ . Denote  $\Psi(v^a, f) =$  $co(R(v^a, f)) \times W(v^a, f)$  for each  $v^a \in V_a$  and  $f \in F$ .

As shown in [35],  $\Psi$  is nonempty, convex, compact valued and upper hemicontinuous. By Fan-Glicksberg's Fixed Point Theorem,  $\Psi$  has a fixed point  $(v^{a'}, f^{l'}) \in V_a \times F$ . Let  $v^{l'}$  be the mapping from  $S_l$  to  $\mathbb{R}^m$  that is generated by  $v^{a'}$  and  $f^{l'}$  through the  $\beta$ -contraction mapping  $\Pi$  as above. Then  $v^{a'}$  is an Smeasurable selection of co $(P(v^{a'}, v^{l'}, \cdot))$  restricted to  $S_a$ ; and furthermore we have for each  $i \in I$  and  $s_l \in S_l$ ,

$$v_i^{l'}(s_l) = \int_{X_{-i}} \int_{X_i} \Phi_i(s_l, x_i, x_{-i}, v^{a'}, v^{l'}) f_i^{l'}(\mathrm{d}x_i|s_l) f_{-i}^{l'}(\mathrm{d}x_{-i}|s_l), \tag{9}$$

$$\Pi_i(v^{a'}, f_{-i}^{l'})(v^{l'})(s_l) = v_i^{l'}(s_l).$$
(10)

Following the same argument as in the proof of Theorem 1, there exists an S-measurable selection  $v^{a*}$  of  $P_{(v^{a'},v^{l'})}$  such that  $\mathbb{E}(v^{a*}\rho_j|\mathcal{G}) = \mathbb{E}(v^{a'}\rho_j|\mathcal{G})$  for each  $1 \leq j \leq J$ , where the conditional expectation is taken on  $(S_a, \mathcal{S}^{S_a}, \lambda^{S_a})$  with  $\lambda^{S_a}$  the normalized probability measure on  $(S_a, \mathcal{S}^{S_a})$ . For any  $s \in S$  and  $x \in A(s)$ ,  $\Phi_i(s, x, v^{a'}, v^{l'}) = \Phi_i(s, x, v^{a*}, v^{l'})$ ,  $\Gamma(v^{a'}, v^{l'}, s) = \Gamma(v^{a*}, v^{l'}, s)$ , and therefore

 $P(v^{a'}, v^{l'}, s) = P(v^{a*}, v^{l'}, s)$ . Thus,  $v^{a*}$  is an  $\mathcal{S}$ -measurable selection of  $P_{(v^{a*}, v^{l'})}$ , and there exists an  $\mathcal{S}$ -measurable mapping  $f^{a*} \colon S_a \to \bigotimes_{i \in I} \mathcal{M}(X_i)$  such that  $f^{a*}(s)$  is a mixed-strategy Nash equilibrium of the game  $\Gamma(v^{a*}, v^{l'}, s)$  and  $v^{a*}(s)$ the corresponding equilibrium payoff.

Let  $v^*(s)$  be  $v^{a*}(s)$  for  $s \in S_a$  and  $v^{l'}(s)$  for  $s \in S_l$ , and  $f^*(s)$  be  $f^{a*}(s)$  for  $s \in S_a$ and  $f^{l'}(s)$  for  $s \in S_l$ . For  $s_a \in S_a$ , since  $v^{a*}$  is a measurable selection of  $P_{(v^{a*},v^{l'})}$  on  $S_a$ , the equilibrium property of  $f^{a*}(s_a)$  then implies that Equations (1) and (2) hold for  $v^*$  and  $f^*$ . Next, for  $s_l \in S_l$ , the identity  $\Phi_i(s_l, x, v^{a'}, v^{l'}) = \Phi_i(s_l, x, v^{a*}, v^{l'})$ implies that Equations (9) and (10) still hold when  $v^{a'}$  is replaced by  $v^{a*}$ , which means that Equations (1) and (2) hold for  $v^*$  and  $f^*$ . Therefore,  $f^*$  is a stationary Markov perfect equilibrium.

### 8.4 Proof of Proposition 2

In this subsection, we shall follow the notations in Subsection 4.1. As discussed in Remark 1, a noisy stochastic game is a stochastic game with endogenous shocks in which  $H^d$  and  $X^d$  are both singletons. As a result, we can slightly abuse the notations by viewing  $Q_R$  to be a mapping from  $H \times \mathscr{R}$  to [0, 1], and its Radon-Nikodym derivative  $\psi$  to be a mapping defined on  $H \times R$ . For simplicity, we denote  $\nu_h(\cdot) = Q_R(\cdot|h)$ .

Let  $\lambda(Z) = \int_H \int_R \mathbf{1}_Z(h, r) \psi(r|h) \, d\nu(r) \, d\kappa(h)$  for all  $Z \in S$ , and  $\mathcal{G} = \mathcal{H} \otimes \{\emptyset, R\}$ . Recall that  $\phi(\cdot|s, x)$  is the Radon-Nikodym derivative of  $Q_H(\cdot|s, x)$  with respect to  $\kappa$ . For each (s, x),  $\phi(\cdot|s, x)$  is a mapping which does not depend on r, and hence is  $\mathcal{G}$ -measurable. We need to show that  $\mathcal{S}$  has no  $\mathcal{G}$ -atom under  $\lambda$ .

Fix any Borel subset  $D \subseteq S$  with  $\lambda(D) > 0$ . There is a measurable mapping  $\alpha$  from  $(D, \mathcal{S}^D)$  to  $(L, \mathcal{B})$  such that  $\alpha$  can generate the  $\sigma$ -algebra  $\mathcal{S}^D$ , where L is the unit interval endowed with the Borel  $\sigma$ -algebra  $\mathcal{B}$ . Let g(h, r) = h for each  $(h, r) \in D$ ,  $D_h = \{r \colon (h, r) \in D\}$  and  $H_D = \{h \in H : \nu_h(D_h) > 0\}$ .

Denote  $g_h(\cdot) = g(h, \cdot)$  and  $\alpha_h(\cdot) = \alpha(h, \cdot)$  for each  $h \in H_D$ . Define a mapping  $f: H_D \times L \to [0, 1]$  as follow:  $f(h, l) = \frac{\nu_h(\alpha_h^{-1}([0, l]))}{\nu_h(D_h)}$ . Similarly, denote  $f_h(\cdot) = f(h, \cdot)$  for each  $h \in H_D$ . For  $\kappa$ -almost all  $h \in H_D$ , the atomlessness of  $\nu_h$  implies  $\nu_h \circ \alpha_h^{-1}(\{l\}) = 0$  for all  $l \in L$ . Thus, the distribution function  $f_h(\cdot)$  is continuous

on L for  $\kappa$ -almost all  $h \in H_D$ .

Let  $\gamma(s) = f(g(s), \alpha(s))$  for each  $s \in D$ , and  $D_0 = \gamma^{-1}([0, \frac{1}{2}])$ , which is a subset of D. For  $h \in H_D$ , let  $l_h$  be max $\{l \in L: f_h(l) \leq \frac{1}{2}\}$  if  $f_h$  is continuous, and 0 otherwise. When  $f_h$  is continuous,  $f_h(l_h) = 1/2$ . For any  $E \in \mathcal{H}$ , let  $D_1 = (E \times R) \cap D$ , and  $E_1 = E \cap H_D$ . If  $\lambda(D_1) = 0$ , then

$$\begin{split} \lambda(D_0 \setminus D_1) &= \lambda(D_0) = \int_{H_D} \nu_h \circ \alpha_h^{-1} \circ f_h^{-1} \left( [0, \frac{1}{2}] \right) \mathrm{d}\kappa(h) \\ &= \int_{H_D} \nu_h \left( \alpha_h^{-1}([0, l_h]) \right) \mathrm{d}\kappa(h) = \int_{H_D} f(h, l_h) \nu_h(D_h) \,\mathrm{d}\kappa(h) \\ &= \frac{1}{2} \int_{H_D} \nu_h(D_h) \,\mathrm{d}\kappa(h) = \frac{1}{2} \lambda(D) > 0. \end{split}$$

If  $\lambda(D_1) > 0$ , then

$$\lambda(D_1 \setminus D_0) = \int_{E_1} \int_R \mathbf{1}_{D \setminus D_0}(h, r) \, \mathrm{d}\nu_h(r) \, \mathrm{d}\kappa(h) = \int_{E_1} \nu_h \circ \alpha_h^{-1} \circ f_h^{-1}\left(\left(\frac{1}{2}, 1\right)\right) \, \mathrm{d}\kappa(h)$$
$$= \int_{E_1} \nu_h \circ \alpha_h^{-1} \circ f_h^{-1}\left([0, 1] \setminus [0, \frac{1}{2}]\right) \, \mathrm{d}\kappa(h) = \frac{1}{2} \int_{E_1} \nu_h(D_h) \, \mathrm{d}\kappa(h) = \frac{1}{2} \lambda(D_1) > 0.$$

Hence, D is not a  $\mathcal{G}$ -atom. Therefore,  $\mathcal{S}$  has no  $\mathcal{G}$ -atom and the condition of coarser transition kernel is satisfied.

### 8.5 **Proof of Proposition 3**

Given the state  $s \in [0, 1)$  and action profile x in the previous stage, suppose that players C, C', D and D' play the strategy 1, the transition probability in the current stage is

$$Q(s,x) = \left(1 - \frac{1}{16}(1-s)\right)\delta_1 + \frac{1}{16}(1-s)U(s,1).$$

It is clear that U(s, 1) is absolute continuous with respect to the Lebesgue measure  $\eta$  with the Radon-Nikodym derivative

$$q(s_1|s) = \begin{cases} \frac{1}{1-s} & s_1 \in [s,1], \\ 0 & s_1 \in [0,s). \end{cases}$$

Suppose that the atomless part Q(s,x)U(s,1) has a decomposable coarser transition kernel; so does U(s,1). Then, for some positive integer J, we have  $q(\cdot|s) = \sum_{1 \leq j \leq J} q_j(\cdot,s)\rho_j(\cdot)$  for any  $s \in [0,1)$ , where  $q_j$  is nonnegative and product measurable,  $\rho_j$  is nonnegative and integrable. Let  $\mathcal{G}$  be the minimal  $\sigma$ -algebra (with strong completion) with respect to which  $q_j(\cdot,s)$  is measurable for all  $1 \leq j \leq J$ and  $s \in [0,1)$ . The condition of a decomposable coarser transition kernel implies that  $\mathcal{B}$  has no  $\mathcal{G}$ -atom. We shall show this condition is not satisfied and thus derive a contradiction.

Denote  $D_j = \{s_1 \in S : \rho_j(s_1) = 0\}$  for  $1 \le j \le J$ . Since  $q(s_1|0) = 1$  for all  $s_1 \in S$ , we must have  $\bigcap_{1 \le j \le J} D_j = \emptyset$ , and hence  $\eta(\bigcap_{1 \le j \le J} D_j) = 0$ .

First suppose that  $\eta(D_j) = 0$  for all j. Let  $\overline{D} = \bigcup_{1 \leq j \leq J} D_j$ ; then  $\eta(\overline{D}) = 0$ . Fix  $s' \in [0,1)$ . Let  $E_j = \{s_1 \in S : q_j(s_1,s') = 0\}$  and  $E_0 = \bigcap_{1 \leq j \leq J} E_j$ . Then  $E_j \in \mathcal{G}$  for  $1 \leq j \leq J$ , and hence  $E_0 \in \mathcal{G}$ . For any  $s_1 \in [s',1]$ , since  $q(s_1|s') > 0$ , there exists  $1 \leq j \leq J$  such that  $q_j(s_1|s') > 0$ , which means that  $s_1 \notin E_j$  and  $s_1 \notin E_0$ . Hence,  $E_0 \subseteq [0,s')$ . For any  $s_1 \in ([0,s') \setminus \overline{D})$ , we have  $q(s_1|s') = 0$ , and  $\rho_j(s_1) > 0$  for each  $1 \leq j \leq J$ , which implies that  $q_j(s_1|s') = 0$  for each  $1 \leq j \leq J$ , and  $s_1 \in E_0$ . That is,  $([0,s') \setminus \overline{D}) \subseteq E_0$ . Hence,  $\eta(E_0 \triangle [0,s']) = 0$ . Therefore,  $[0,s'] \in \mathcal{G}$  for all  $s' \in [0,1)$ , which implies that  $\mathcal{G}$  coincides with  $\mathcal{B}$  and  $\mathcal{B}$  has a  $\mathcal{G}$ -atom [0,1). This is a contradiction.

Next suppose that  $\eta(D_j) = 0$  does not hold for all j. Then there exists a set, say  $D_1$ , such that  $\eta(D_1) > 0$ . Let  $Z = \{K \subseteq \{1, \ldots, J\} \colon 1 \in K, \eta(D^K) > 0\}$ , where  $D^K = \cap_{j \in K} D_j$ . Hence,  $\{1\} \in Z, Z$  is finite and nonempty. Let  $K_0$  be the element in Z containing most integers; that is,  $|K_0| \ge |K|$  for any  $K \in Z$ , where |K| is the cardinality of K. Let  $K_0^c = \{1, \ldots, J\} \setminus K_0$ . Then  $K_0^c$  is not empty since  $\eta(\cap_{1 \le j \le J} D_j) = 0$ . In addition,  $\eta(D^{K_0} \cap D_j) = 0$  for any  $j \in K_0^c$ . Otherwise,  $\eta(D^{K_0} \cap D_j) > 0$  for some  $j \in K_0^c$  and hence  $(K_0 \cup \{j\}) \in Z$ , which contradicts the choice of  $K_0$ . Let  $\hat{D} = \bigcup_{k \in K_0^c} (D^{K_0} \cap D_k)$ ; then  $\eta(\hat{D}) = 0$ . For all  $s_1 \in D^{K_0}$ ,  $q(s_1|s) = \sum_{k \in K_0^c} q_k(s_1, s)\rho_k(s_1)$  for all  $s \in [0, 1)$ .

Fix  $s' \in [0, 1)$ . Let  $E_k = \{s_1 \in S : q_k(s_1, s') = 0\}$  and  $E_{K_0^c} = \bigcap_{k \in K_0^c} E_k$ . Then  $E_k \in \mathcal{G}$  for any k and hence  $E_{K_0^c} \in \mathcal{G}$ . For any  $s_1 \in [s', 1]$ , since  $q(s_1|s') > 0$ , there exists  $k \in K_0^c$  such that  $q_k(s_1|s') > 0$ , which means that  $s_1 \notin E_k$  and

 $s_1 \notin E_{K_0^c}$ . Hence,  $E_{K_0^c} \subseteq [0, s')$ , and  $E_{K_0^c} \cap D^{K_0} \subseteq [0, s') \cap D^{K_0}$ . Now, for any  $s_1 \in \left(\left([0, s') \cap D^{K_0}\right) \setminus \hat{D}\right)$ , we have  $q(s_1|s') = 0$ , and  $\rho_k(s_1) > 0$  for each  $k \in K_0^c$ , which implies that  $q_k(s_1|s') = 0$  for each  $k \in K_0^c$ , and  $s_1 \in E_{K_0^c}$ . That is,  $\left(\left([0, s') \cap D^{K_0}\right) \setminus \hat{D}\right) \subseteq E_{K_0^c} \cap D^{K_0}$ . Hence,  $([0, s'] \cap D^{K_0}) \setminus (E_{K_0^c} \cap D^{K_0}) \subseteq \hat{D}$ , and  $\eta\left((E_{K_0^c} \cap D^{K_0}) \Delta([0, s'] \cap D^{K_0})\right) = 0$ . Thus,  $\mathcal{B}$  has a  $\mathcal{G}$ -atom  $D^{K_0}$ . This is again a contradiction.

#### 8.6 Proof of Propositions 4 and 5

Proof of Proposition 4. Define a correspondence

$$G(s) = \begin{cases} \{0,1\} & s \in D; \\ \{0\} & s \notin D. \end{cases}$$

We claim that  $\mathcal{I}_{G}^{(\mathcal{S},\mathcal{G},\lambda)} \neq \mathcal{I}_{\mathrm{CO}(G)}^{(\mathcal{S},\mathcal{G},\lambda)}$ . Let  $g_{1}(s) = \frac{1}{2}\mathbf{1}_{D}$ , where  $\mathbf{1}_{D}$  is the indicator function of the set D. Then  $g_{1}$  is an  $\mathcal{S}$ -measurable selection of  $\mathrm{co}(G)$ . If there is an  $\mathcal{S}$ -measurable selection  $g_{2}$  of G such that  $\mathbb{E}^{\lambda}(g_{1}|\mathcal{G}) = \mathbb{E}^{\lambda}(g_{2}|\mathcal{G})$ , then there is a subset  $D_{2} \subseteq D$  such that  $g_{2}(s) = \mathbf{1}_{D_{2}}$ . Since D is a  $\mathcal{G}$ -atom, for any  $\mathcal{S}$ -measurable subset  $E \subseteq D$ , there is a subset  $E_{1} \in \mathcal{G}$  such that  $\lambda(E \triangle (E_{1} \cap D)) = 0$ . Then

$$\lambda(E \cap D_2) = \int_S \mathbf{1}_E(s)g_2(s) \,\mathrm{d}\lambda(s) = \int_S \mathbb{E}^\lambda (\mathbf{1}_{E_1}\mathbf{1}_D g_2|\mathcal{G}) \,\mathrm{d}\lambda = \int_S \mathbf{1}_{E_1} \mathbb{E}^\lambda (g_2|\mathcal{G}) \,\mathrm{d}\lambda$$
$$= \int_S \mathbf{1}_{E_1} \mathbb{E}^\lambda (g_1|\mathcal{G}) \,\mathrm{d}\lambda = \frac{1}{2} \int_S \mathbf{1}_{E_1}\mathbf{1}_D \,\mathrm{d}\lambda = \frac{1}{2}\lambda(E_1 \cap D) = \frac{1}{2}\lambda(E).$$

Thus,  $\lambda(D_2) = \frac{1}{2}\lambda(D) > 0$  by choosing E = D. However,  $\lambda(D_2) = \frac{1}{2}\lambda(D_2)$  by choosing  $E = D_2$ , which implies that  $\lambda(D_2) = 0$ . This is a contradiction.

Proof of Proposition 5. Suppose that there exists an  $\mathcal{S}$ -measurable selection g of G such that  $\mathbb{E}^{\lambda}(g\rho_n|\mathcal{F}) = 0$  for any  $n \geq 0$ . Then there exists a set  $E \in \mathcal{S}$  such that

$$g(s) = \begin{cases} 1 & s \in E; \\ -1 & s \notin E. \end{cases}$$

Thus,

$$\lambda(E) - \lambda(E^c) = \int_S g\rho_0 \,\mathrm{d}\lambda = \int_S \mathbb{E}^\lambda(g\rho_0|\mathcal{F}) \,\mathrm{d}\lambda = 0,$$

which implies  $\lambda(E) = \frac{1}{2}$ . Moreover,

$$\int_{S} g\rho_n \, \mathrm{d}\lambda = \int_{S} g\rho_n \, \mathrm{d}\lambda - \int_{S} g \, \mathrm{d}\lambda = \int_{S} \mathbb{E}^{\lambda}(g\rho_n | \mathcal{F}) \, \mathrm{d}\lambda - 0 = 0$$

for each  $n \ge 1$ , which contradicts with the condition that  $\{\varrho_n\}_{n\ge 0}$  is a complete orthogonal system.

### References

- C. D. Aliprantis and K. C. Border, *Infinite Dimensional Analysis: A Hitchhiker's Guide*, Springer, Berlin, 2006.
- [2] R. Amir, Continuous Stochastic Games of Capital Accumulation with Convex Transitions, Games and Economic Behavior 15 (1996), 111–131.
- [3] R. Amir, Complementarity and Diagonal Dominance in Discounted Stochastic Games, Annals of Operations Research 114 (2002), 39–56.
- [4] R. Amir, Stochastic Games in Economics and Related Fields: An Overview, in *Stochastic Games and Applications* (A. Neyman and S. Sorin ed.), NATO Adv. Sci. I. C-MAT, Vol. 570, Kluwer Academic, Dordrecht, 2003, 455–470.
- [5] L. Balbus, K. Reffett and L. Woźny, A Constructive Study of Markov Equilibria in Stochastic Games with Strategic Complementarities, *Journal* of Economic Theory 150 (2014), 815–840.
- [6] L. Balbus and A. S. Nowak, Existence of Perfect Equilibria in A Class of Multigenerational Stochastic Games of Capital Accumulation, *Automatica* 44 (2008), 1471–1479.
- [7] D. P. Bertsekas and S. E. Shreve, Stochastic Optimal Control: The Discrete Time Case, New York, Academic Press, 1978.
- [8] D. Blackwell, Discounted Dynamic Programming, The Annals of Mathematical Statistics 36 (1965), 226–235.
- [9] L. O. Curtat, Markov Equilibria of Stochastic Games with Complementarities, Games and Economic Behavior 17 (1996), 177–199.
- [10] D. Duffie, J. Geanakoplos, A. Mas-Colell and A. McLennan, Stationary Markov Equilibria, *Econometrica* 62 (1994), 745–781.
- [11] J. Duggan, Noisy Stochastic Games, *Econometrica* 80 (2012), 2017–2045.
- [12] E. B. Dynkin and I. V. Evstigneev, Regular Conditional Expectations of Correspondences, *Theory of Probability and its Applications* **21** (1977), 325– 338.

- [13] R. Ericson and A. Pakes, Markov-perfect Industry Dynamics: A Framework for Empirical Work, *Review of Economic Studies* 62 (1995), 53–82.
- [14] A. Federgruen, On n-Person Stochastic Games with Denumerable State Space, Advances in Applied Probability 10 (1978), 452–471.
- [15] A. Fink, Equilibrium in a Stochastic n-Person Game, Journal of Science of the Hiroshima University 28 (1964), 89–93.
- [16] I. L. Glicksberg, A Further Generalization of the Kakutani Fixed Point Theorem with Application to Nash Equilibrium Points, *Proceedings of the American Mathematical Society* 3 (1952), 170–174.
- [17] A. Hanen and J. Neveu, Atomes conditionnels d'un espace de probabilité, Acta Mathematica Hungarica 17 (1966), 443–449.
- [18] W. He, X. Sun and Y. N. Sun, Modeling Infinitely Many Agents, *Theoretical Economics*, forthcoming, 2016.
- [19] C. J. Himmelberg, T. Parthasarathy, T. E. S. Raghavan and F. S. Van Vleck, Existence of p-Equilibrium and Optimal Stationary Strategies in Stochastic Games, *Proceedings of the American Mathematical Society* **60** (1976), 245– 251.
- [20] H. Hopenhayn, Entry, Exit, and Firm Dynamics in Long Run Equilibrium, Econometrica 60 (1992), 1127–1150.
- [21] U. Horst, Stationary Equilibria in Discounted Stochastic Games with Weakly Interacting Players, Games and Economic Behavior 51 (2005), 83–108.
- [22] A. Jaśkiewicz and A. S. Nowak, Stationary Markov Perfect Equilibria in Risk Sensitive Stochastic Overlapping Generations Models, *Journal of Economic Theory* 151 (2014), 411–447.
- [23] A. Jaśkiewicz and A. S. Nowak, Stochastic Games of Resource Extraction, Automatica 54 (2015), 310–316.
- [24] A. Jaśkiewicz and A. S. Nowak, Zero-Sum Stochastic Games, Manuscript (2016a).
- [25] A. Jaśkiewicz and A. S. Nowak, Non-Zero-Sum Stochastic Games, Manuscript (2016b).
- [26] K. Jacobs, *Measure and Integral*, Edited by Z. W. Birnbaum and E. Lukacs, Academic Press, 1978.
- [27] Y. Levy, A Discounted Stochastic Game with No Stationary Nash Equilibrium: Two Examples, *Econometrica* 81 (2013), 1973–2007.
- [28] Y. Levy and A. McLennan, Corrigendum to: "Discounted Stochastic Games with No Stationary Nash Equilibrium: Two Examples", *Econometrica* 83 (2015), 1237–1252.
- [29] E. Maskin and J. Tirole, A Theory of Dynamic Oligopoly, I: Overview and Quantity Competition with Large Fixed Costs, *Econometrica* 56 (1988), 549–569.

- [30] E. Maskin and J. Tirole, A Theory of Dynamic Oligopoly, II: Price Competition, Kinked Demand Curves, and Edgeworth Cycles, *Econometrica* 56 (1988), 571–599.
- [31] E. Maskin and J. Tirole, Markov perfect equilibrium: I. Observable actions, Journal of Economic Theory 100 (2001), 191–219.
- [32] J.-F. Mertens and T. Parthasarathy, Equilibria for Discounted Stochastic Games, CORE Discussion Paper No. 8750, Universitè Catholique de Louvain, 1987.
- [33] J. Neveu, Atomes Conditionnels d'Espaces de Probalite et Theorie de l'Information, Symposium on Probability Methods in Analysis, Springer, Berlin Heidelberg, 1967, 256–271.
- [34] A. Neyman and S. Sorin, eds. Stochastic Games and Applications, NATO Adv. Sci. I. C-MAT, Vol. 570, Kluwer Academic, Dordrecht, 2003.
- [35] A. S. Nowak, On a New Class of Nonzero-Sum Discounted Stochastic Games Having Stationary Nash Equilibrium Points, *International Journal of Game Theory* **32** (2003), 121–132.
- [36] A. S. Nowak, On Perfect Equilibria in Stochastic Models of Growth with Intergenerational Altruism, *Economic Theory* 28 (2006), 73–83.
- [37] A. S. Nowak, On Stochastic Games in Economics, Mathematical Methods of Operations Research 66 (2007), 513–530.
- [38] A. S. Nowak and T. E. S. Raghavan, Existence of Stationary Correlated Equilibria with Symmetric Information for Discounted Stochastic Games, *Mathematics of Operations Research* 17 (1992), 519–527.
- [39] T. Parthasarathy, Discounted Positive, Noncooperative Stochastic Games, International Journal of Game Theory 2 (1973), 25–37.
- [40] T. Parthasarathy and S. Sinha, Existence of Stationary Equilibrium Strategies in Non-Zero Sum Discounted Stochastic Games with Uncountable State Space and State-Independent Transitions, *International Journal of Game Theory* 18 (1989), 189–194.
- [41] P. Rogers, Non-Zero Sum Stochastic Games, Operations Research Center Report 69–8, University of California, Berkeley, 1969.
- [42] L. Shapley, Stochastic Games, Proceedings of the National Academy of Sciences USA 39 (1953), 1095–1100.
- [43] M. Sobel, Non-Cooperative Stochastic Games, Annals of Mathematical Statistics 42 (1971), 1930–1935.
- [44] M. Takahashi, Equilibrium Points of Stochastic Non-Cooperative n-Person Games, Journal of Science of the Hiroshima University 28 (1964), 95–99.
- [45] W. Whitt, Representation and Approximation of Noncooperative Sequential Games, SIAM Journal of Control and Optimization 18 (1980), 33–48.